LOAN PORTFOLIO SALE AND PURCHASE AGREEMENT

dated as of July 16, 2008

among

AMERITRANS CAPITAL CORPORATION,

ELK ASSOCIATES FUNDING CORPORATION,

MEDALLION BANK

AND

MEDALLION FINANCIAL CORP.

TABLE OF CONTENTS

Page

ARTICLE 1	DEFINITIONS	1	
ARTICLE 2	PURCHASE AND SALE OF LOAN PORTFOLIO	6	
2.1	Purchase and Sale of Loan Portfolio; Excluded Loans; Assumed Obligations; Excluded Obligations; Schedule of Acquired Loans; Obligations of the Parties		
2.2	Closing	9	
2.3	Purchase Price; Payment at Closing; Deposit; Post-Closing Adjustments to Purchase Price; Adjustments and Allocations with respect to Payments on Acquired Loans	9	
2.4	Assignments and Endorsements	12	
2.5	Survival of Responsibilities	14	
2.6	Reimbursement for Use of Seller's Employees	14	
2.7	True Sale; Security Agreement	15	
ARTICLE 3	REPRESENTATIONS AND WARRANTIES OF SELLER PARTIES	15	
3.1	Seller Parties' Representations and Warranties	15	
3.2	Disclaimers	20	
ARTICLE 4	REPRESENTATIONS AND WARRANTIES OF BUYER PARTIES	20	
4.1	Buyer Parties' Representations and Warranties	20	
ARTICLE 5	COVENANTS OF SELLER PARTIES AND BUYER PARTIES	21	
5.1	General	21	
5.2	Proxy Statement	21	
5.3	Governmental and Third-Party Consents	22	
5.4	Tax Matters	22	
5.5	Confidentiality	23	
5.6	Non-Competition	23	
5.7	Conduct of Business Prior to Closing	25	
5.8	Expenses	25	
5.9	Additional Seller's Covenants to Buyer	25	
5.10	Additional Buyer's Covenants to Seller	27	
ARTICLE 6	CONDITIONS TO CLOSING AND TERMINATION	27	

TABLE OF CONTENTS (continued)

Page

6.1	Conditions to Closing	27
6.2	Termination	29
ARTICLE 7	BUYER PRE-CLOSING REMEDY	30
7.1	Pre-Closing Remedy	
ARTICLE 8	INDEMNIFICATION	30
8.1	Seller Parties' Indemnification Obligations	
8.2	Buyer Parties' Indemnification Obligations	31
8.3	Obligation to Mitigate Losses	31
8.4	Tax Consequences of Indemnification	31
8.5	Procedures	31
8.6	Limitations on Indemnification	
8.7	Survival of Indemnification Obligations	33
8.8	Exclusive Remedy	33
ARTICLE 9	MISCELLANEOUS PROVISIONS	
/ INTICLE /		
9.1	Notices	
		33
9.1	Notices	33 34
9.1 9.2	Notices Severability	33 34 34
9.1 9.2 9.3	Notices Severability Governing Law	33 34 34 34
9.1 9.2 9.3 9.4	Notices Severability Governing Law Assignments; Binding Effect	33 34 34 34 34
9.19.29.39.49.5	Notices Severability Governing Law Assignments; Binding Effect Survival	33 34 34 34 34 35
 9.1 9.2 9.3 9.4 9.5 9.6 	Notices Severability Governing Law Assignments; Binding Effect Survival Entire Agreement; Amendments	33 34 34 34 35 35
 9.1 9.2 9.3 9.4 9.5 9.6 9.7 	Notices	33 34 34 34 35 35
 9.1 9.2 9.3 9.4 9.5 9.6 9.7 9.8 	Notices Severability Governing Law Assignments; Binding Effect Survival Entire Agreement; Amendments Counterparts; Facsimile; PDF	33 34 34 34 35 35 35
 9.1 9.2 9.3 9.4 9.5 9.6 9.7 9.8 9.9 	Notices Severability Governing Law Assignments; Binding Effect Survival Entire Agreement; Amendments Counterparts; Facsimile; PDF Announcements Waiver of Jury Trial	33 34 34 34 35 35 35 35 35
 9.1 9.2 9.3 9.4 9.5 9.6 9.7 9.8 9.9 9.10 	Notices	33 34 34 35 35 35 35 35 35 35 35
 9.1 9.2 9.3 9.4 9.5 9.6 9.7 9.8 9.9 9.10 9.11 	Notices Severability Governing Law Assignments; Binding Effect Survival Entire Agreement; Amendments Counterparts; Facsimile; PDF Announcements Waiver of Jury Trial Consent to Jurisdiction No Third Party Beneficiary	33 34 34 34 35 35 35 35 35 35 36 36

LOAN PORTFOLIO SALE AND PURCHASE AGREEMENT (the "<u>Agreement</u>"), dated as of July 16, 2008, by and among Ameritrans Capital Corporation, a Delaware corporation ("<u>Ameritrans</u>"), Elk Associates Funding Corporation ("<u>Seller</u>"), a New York corporation licensed as a Small Business Investment Company by the United States Small Business Administration (the "<u>SBA</u>") and a wholly owned subsidiary of Ameritrans, Medallion Financial Corp., a Delaware corporation ("<u>MFC</u>") and Medallion Bank, a Utah industrial bank and a wholly owned subsidiary of MFC ("<u>Buyer</u>").

PRELIMINARY STATEMENTS

Capitalized terms used herein have the meanings set forth or referred to in Article 1 hereof.

Seller owns the Loan Portfolio and desires to sell and transfer the Loan Portfolio to Buyer, and Buyer desires to purchase the Loan Portfolio.

Accordingly, in consideration of the mutual covenants herein contained, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties do hereby agree as follows:

ARTICLE 1

DEFINITIONS

For purposes of this Agreement, the following terms shall have the meanings indicated below:

"<u>Acquired Loan</u>" means each loan made by Seller that is a Medallion Loan.

"<u>Adjustment Schedule</u>" has the meaning given such term in Section 2.3.4(b) of this Agreement.

"<u>Affiliate</u>" means, with respect to any Person, any other Person that directly or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such Person.

"<u>Agreement</u>" means this Loan Portfolio Sale and Purchase Agreement, among Ameritrans, Seller, on the one hand, and MFC and Buyer, on the other hand, including all Schedules and Exhibits attached hereto and referenced herein and any written amendments hereto.

"<u>Allocated Purchase Price</u>" means, with respect to each Acquired Loan, the portion of the Purchase Price allocated to such Acquired Loan in accordance with Section 2.3.2(a) of this Agreement.

"<u>Allocation</u>" has the meaning given such term in Section 5.4.1 of this Agreement.

"<u>Ameritrans</u>" has the meaning given such term in the preamble to this Agreement.

"<u>Ameritrans Stockholders Meeting</u>" has the meaning given such term in Section 5.9.2 of this Agreement.

"<u>Assumed Obligations</u>" has the meaning given such term in Section 2.1.3(a) of this Agreement.

"<u>Business Day</u>" means any day other than a Saturday, Sunday or a day on which banking institutions in New York, New York or Salt Lake City, Utah are authorized or obligated by law or executive order to be closed.

"<u>Buyer</u>" has the meaning given such term in the preamble to this Agreement.

"<u>Buyer Indemnified Party</u>" and "<u>Buyer Indemnified Parties</u>" have the meanings given such term in Section 8.1.1 of this Agreement.

"Buyer Parties" means Buyer and MFC.

"<u>Closing</u>" means the payment of the Purchase Price for the Loan Portfolio in accordance with Section 2.3 of this Agreement by Buyer and the delivery of the Closing Documents by Buyer and Seller.

"<u>Closing Date</u>" means the second Business Day after the satisfaction or waiver of the last to be satisfied or waived of the conditions set forth in Article 6 of this Agreement (other than those conditions that by their terms are to be satisfied at the Closing), which is the date on which the Closing occurs.

"<u>Closing Date Payment Amount</u>" has the meaning given such term in Section 2.3.2(b) of this Agreement.

"<u>Closing Documents</u>" means the documents required to be delivered under the terms of this Agreement at the Closing by Seller and Buyer, including (i) the bill of sale, (ii) any notes endorsed by Seller as set forth in Section 2.4.4, and (iii) any further assignments or other documents as Buyer or Seller deem necessary or appropriate pursuant to this Agreement for the legal transfer to Buyer of Seller's right, title and interest in and to the Loan Portfolio, all of which must be in form and substance reasonably acceptable to Seller and Buyer.

"Code" means the Internal Revenue Code of 1986, as amended.

"<u>Collateral</u>" means, with respect to any Acquired Loan, any personal or real property securing such Acquired Loan under the Loan Agreement or related Loan Documents for such Acquired Loan, including, without limitation, any taxicab medallions and related assets.

"<u>Confidentiality Agreement</u>" has the meaning given such term in Section 5.5 of this Agreement.

"<u>Deposit</u>" has the meaning given such term in Section 2.3.3 of this Agreement.

"<u>Encumbrances</u>" means any mortgage, security interest, lien, pledge, claim, charge, any restriction on the transfer of any security or other asset, any restriction on the receipt of any income derived from any asset, any restriction on the use of any asset and any restriction on the possession, exercise or transfer of any attribute of ownership of any asset, or other encumbrance.

"Escrow Agent" means Stursberg and Associates, LLC.

"<u>Escrow Agreement</u>" means the Escrow Agreement, dated as of the date hereof, by and among the Escrow Agent, Buyer and Seller.

"<u>Excluded Loans</u>" has the meaning given such term in Section 2.1.2 of this Agreement.

"<u>Excluded Obligations</u>" has the meaning given such term in Section 2.1.3(b) of this Agreement.

"<u>Final Closing Date Statement</u>" has the meaning given such term in Section 2.3.4(a) of this Agreement.

"<u>Final Purchase Price</u>" has the meaning given such term in Section 2.3.4(a) of this Agreement.

"<u>Governmental Entity</u>" means a federal, state or other governmental, regulatory or administrative agency, court, department, commission, board, bureau, or other authority or instrumentality, domestic or foreign.

"<u>Granoff Employment Agreement</u>" has the meaning given such term in Section 5.6.3 of this Agreement.

"<u>Indemnified Party</u>" means any Buyer Indemnified Party or Seller Indemnified Party.

"<u>Indemnifying Party</u>" has the meaning given such term in Section 8.5.1 of this Agreement.

"<u>Independent Auditor</u>" has the meaning given such term in Section 2.3.4(a) of this Agreement.

"<u>Knowledge</u>" means (a) with respect to any Seller Party, the actual knowledge of the officers and employees of such Seller Party, and (b) with respect to any Buyer Party, the actual knowledge of the officers and employees of such Buyer Party, in each case after due and reasonable inquiry.

"Loan Agreement" means, with respect to any Acquired Loan, the loan agreement (or loan and security agreement) relating to such Acquired Loan.

"Loan Documents" means, with respect to any Acquired Loan, (a) the Loan Agreement for such Acquired Loan and any and all promissory notes, collateral security agreements, mortgages, UCC-1 financing statements and evidence of ownership of a taxi medallion from the applicable regulatory authority and, to the extent applicable to any Acquired Loan, (b) any guaranties, assignments, pledges, cross-default and cross-collateralization agreements, lien priority agreements, intercreditor agreements, participation agreements, broking arrangements, servicing arrangements, and any other agreements, documents or instruments evidencing a payment obligation under, or providing security or any other credit enhancement for, such Acquired Loan, and all amendments, modifications, renewals, extensions, rearrangements and substitutions with respect to any of the foregoing.

"Loan Files" means, with respect to any Acquired Loan, the Loan Documents and all other credit, documentation and transaction files of Seller relating to such Acquired Loan, which files (a) shall include all credit files, disbursement letters, SBA forms, correspondence and (b) may include general credit information, credit records, customer identification, tax returns (to the extent received from Obligors), payment histories, lien searches, appraisals, property insurance policies, evidence of OFAC compliance, title insurance policies or attorney's title opinions, and other related records or documents, but shall exclude Seller's privileged or confidential communications between Seller and its legal counsel.

"<u>Loan Portfolio</u>" means, collectively, (i) each Acquired Loan, (ii) the Loan Documents, (iii) the Collateral, (iv) the Loan Files, and (v) all other rights, title and interests of Seller in and to the Acquired Loans and related Loan Files.

"<u>Losses</u>" means all losses, causes of action, liabilities, claims, demands, obligations, damages, penalties, fines, forfeitures, judgments, legal fees and other costs, fees and expenses.

"<u>LTV</u>" has the meaning given such term in Section 2.1.2(d) of this Agreement.

"<u>MFC</u>" has the meaning given such term in the preamble to this Agreement.

"<u>Medallion Loan</u>" means each loan made by Seller and identified on Schedule A hereto, as such schedule may be amended pursuant to Section 2.1.4 of this Agreement.

"<u>Objection Period</u>" has the meaning given such term in Section 2.3.4(a) of this Agreement.

"<u>Obligor</u>" means each Person who is now or hereafter liable for the full or partial payment or performance of any Acquired Loan, whether such obligation is direct, indirect, primary, secondary, joint or several.

"<u>Parties</u>" means Buyer and MFC, on the one hand, and Seller and Ameritrans, on the other hand, as parties to this Agreement.

"Person" means, any individual, corporation (including non-profit corporation), general or limited partnership, limited liability company, joint venture,

association, joint stock company, trust, estate, unincorporated organization, government or agency or political subdivision thereof or other entity.

"<u>Preliminary Closing Date Statement</u>" has the meaning given such term in Section 2.3.2(a) of this Agreement.

"<u>Proxy Statement</u>" has the meaning given such term in Section 5.2.1 of this Agreement.

"<u>Repurchase Objection Period</u>" has the meaning given such term in Section 2.3.4(b) of this Agreement.

"<u>Repurchase Objection Statement</u>" has the meaning given such term in Section 2.3.4(b) of this Agreement.

"<u>Required Stockholder Vote</u>" has the meaning given such term in Section 3.1.10 of this Agreement.

"<u>Restricted Activity</u>" means, directly or indirectly, the business of making or purchasing loans or otherwise extending credit related to the financing of taxicab medallions or related assets in the Restricted Markets; <u>provided</u>, <u>however</u>, that this definition shall expressly exclude the brokering of loans solely to Buyer and for which the Persons liable for the full or partial payment or performance of such loans are not Obligors.

"<u>Restricted Markets</u>" means any of (a) Chicago, Illinois; (b) Miami, Florida; (c) Boston/Cambridge, Massachusetts; (d) Newark, New Jersey; and (e) New York, New York.

"<u>SBA</u>" has the meaning given such term in the preamble to this Agreement.

"SEC" means the United States Securities and Exchange Commission.

"<u>Seller</u>" has the meaning given such term in the preamble to this Agreement.

"<u>Seller Indemnified Party</u>" and "<u>Seller Indemnified Parties</u>" have the meanings given such term in Section 8.2.1 of this Agreement.

"Seller Parties" means Seller and Ameritrans.

"<u>Seller Stockholders Meeting</u>" has the meaning given such term in Section 5.9.2 of this Agreement.

"<u>Specified Diversified Loan Event</u>" means the earlier to occur of (a) the closing of the purchase by Buyer from Seller of each loan identified on Schedule B hereto for which the entire unpaid principal amount is not repaid after the date hereof or (b) the repayment of the entire unpaid principal amounts of each loan identified on Schedule B hereto.

"<u>Specified Diversified Loan Agreement</u>" means that certain Specified Diversified Loan Portfolio Sale And Purchase Agreement to be entered into by Ameritrans, Seller, MFC and Freshstart Venture Capital Corp., a wholly owned subsidiary of MFC.

"Tax" or "Taxes" means any federal, state, local or foreign taxes, including, but not limited to, taxes on or measured by income, estimated income, alternative or add-on minimum tax, gross receipts, sales, use, ad valorum, franchise, capital stock, transfer, gains, profits, license, payroll, employment, withholding, social security (or similar), occupation, unemployment, disability, excise, severance, stamp, premium, value added, real property, personal property, inventory toll, duty and any related charge, amount and similar tax, together with any interest, penalty, thereto.

"<u>Tax Return</u>" means any return (including any information return), report, statement, declaration, estimate, schedule, notice, notification, form, election, certificate or other document or information that is, has been or may in the future be filed with or submitted to, or required to be filed with or submitted to, any Governmental Entity in connection with the determination, assessment, collection or payment of any Tax or in connection with the administration, implementation or enforcement of or compliance with any applicable law relating to any Tax.

"<u>Trigger Amount</u>" has the meaning given such term in Section 8.6.1 of this Agreement.

"<u>Trigger Event</u>" has the meaning given such term in Section 5.6.3 of this Agreement.

ARTICLE 2

PURCHASE AND SALE OF LOAN PORTFOLIO

2.1 <u>Purchase and Sale of Loan Portfolio; Excluded Loans; Assumed Obligations;</u> <u>Excluded Obligations; Schedule of Acquired Loans; Obligations of the Parties</u>. In each case, on the terms and subject to the conditions of this Agreement:

2.1.1 <u>Purchase and Sale of Loan Portfolio</u>. Seller hereby agrees to sell to Buyer, and Buyer hereby agrees to purchase from Seller, on the Closing Date for the consideration specified in Section 2.3 below, all of Seller's right, title and interest in and to the Loan Portfolio. The acquired Loan Portfolio shall not include Excluded Loans. Except as set forth in Section 2.3.4(b) hereof, Seller shall not be obligated to repurchase any loans acquired by Buyer on the Closing Date and later determined to meet the qualifications for an Excluded Loan set forth in Section 2.1.2 of this Agreement. For the purpose of clarity, it is the intention of Buyer and Seller that, other than loans repurchased by Seller in accordance with Section 2.3.4(b) hereof, indemnification under Section 8 of this Agreement for breach of representations and warranties shall be Buyer's sole remedy for the inclusion of any loan in the acquired Loan Portfolio that is determined, after the Closing Date, by Buyer or Seller to meet the qualifications for an Excluded Loan set forth in Section 2.1.2 of this Agreement. 2.1.2 <u>Excluded Loans</u>. The Parties agree that Seller is not selling, assigning, transferring, conveying or delivering to Buyer, and the acquired Loan Portfolio shall not include (collectively, the "<u>Excluded Loans</u>"),

(a) Loans which are 60 days or more past due as of the Closing Date and any other loans held by the same or affiliated Obligor or guarantor;

(b) Loans made to any entity where any Affiliate of Buyer is an officer, director or manager of the borrowing entity;

(c) Loans that constitute lines of credit;

(d) Loans for which Seller does not have a perfected first priority security interest in the collateral (except for second loans to the same Obligor which, when the principal balance of the second loan is added to the amount of the first loan, has an aggregate loan to value ratio ("<u>LTV</u>") of less than 80% at the time of Closing);

(e) Loans for which during the twelve months prior to the Closing Date Seller has granted the Obligor an extension, deferral or modification not in accordance with Seller's then existing policies and procedures;

(f) Except as set forth on Schedule 2.1.2(f)(i) hereto, loans with a LTV of greater than 80% of the current market value for such loan at the time of Closing, based upon the methodology and accounting principles employed in preparing Schedule 2.1.2(f)(ii) hereto; <u>provided</u> that Buyer and Seller shall execute a joint participation agreement with respect to each such loan set forth on Schedule 2.1.2(f)(i) hereto pursuant to which Buyer shall be entitled to an 80% senior participating percentage interest and Seller shall be entitled to a 20% junior participating percentage interest, each such joint participation agreement to be in the form of Exhibit B hereto;

(g) Loans relating to which there is pending litigation against either Seller or its agents;

(h) Loans for which the Obligor has filed bankruptcy or is the debtor in a voluntary or involuntary bankruptcy proceeding, or is the subject of a comparable receivership or insolvency proceeding prior to the Closing Date;

(i) Loans with corporate entities which were not duly incorporated at the time of origination;

(j) Loans with respect to which any Obligor is, to the Knowledge of Seller, asserting as of the Closing Date a right of rescission, setoff, counterclaim or defense;

(k) Loans for which the (i) original copies of the Loan Agreement, note, security agreement or, if applicable, guaranty or (ii) original filed UCC-1 or certified filed UCC-1, or other proof of UCC filing and evidence of medallion ownership from the applicable regulatory authority are not available, except in the event that Seller does not have, or is unable to locate, the original promissory note(s) with respect to any Loan Agreement, Seller may deliver to Buyer on the Closing Date, for each such missing original promissory note, either (a) the lost note affidavit or equivalent received by Seller from the transferor with respect thereto, or (b) a separate lost note affidavit prepared and executed by Seller (in such form as Seller may reasonably deem appropriate, including a statement to the effect that Seller has been unable to locate, after the exercise of reasonable diligence, the applicable original promissory note, and without any further representation or recourse to Seller); <u>provided</u>, <u>however</u>, that the principal amount of such missing original promissory notes (including promissory notes for loans originated between the date hereof and the Closing Date) for which Seller delivers to Buyer the items described in (a) and (b) above shall not, in the aggregate, exceed \$500,000;

(1) Any loan which, at the time of origination of such loan, if the Obligor for such loan was an individual, such individual failed to furnish documentation to the applicable taxi commission or regulatory authority evidencing United States citizenship or status as a resident alien with a "Green Card", to provide a social security card, a drivers license or otherwise failed to comply with all requirements of the applicable taxi commission or regulatory authority's rules and regulations at the time of the issuance of the taxi license or medallion to such Obligor by such taxi commission or regulatory authority; provided that the issuance by the applicable taxi commission or regulatory authority of the taxi license or medallion to such individual shall be deemed conclusive proof that such individual complied with the applicable taxi commission's or regulatory authority's requirements;

(m) Any loan originated by Seller for any purpose other than the financing of the applicable Obligor's medallion, and such loan was made other than in the ordinary course of Seller's business;

(n) Any loan for which promissory notes, collateral security agreements, guaranties, UCC-1 filings and evidence of ownership of a taxi medallion from the applicable regulatory authority pertaining to such loan were not fully and properly executed by the parties thereto;

(o) Any loan which, as of the date hereof, the name of the original Obligor of such loan appeared on the Specially Designated Nationals List of the Office of Foreign Assets Control of the United States Department of the Treasury;

(p) Any loan which, as of the Closing Date, carried an interest rate below the greater of (1) the market interest rate and (2) the prime rate of interest; <u>provided</u>, <u>however</u>, that any such loan shall constitute an "Excluded Loan" only to the extent that (X) the principal value of all loans which, as of the Closing Date, carried an interest rate below the greater of clauses (1) and (2) above, exceed \$2,000,000 in the aggregate and (Y) such loan has a maturity date of greater than twelve (12) months from the date hereof; and

(q) Any loan for which the Obligor for such loan (1) is not currently the subject of a judgment in favor of Seller and (2) did not have its related medallion, the Collateral for such loan, foreclosed (or subject to foreclosure).

2.1.3 <u>Assumption of Obligations</u>.

(a) <u>Assumed Obligations</u>. Buyer shall assume on the Closing Date, and shall on and after the Closing Date perform and discharge in accordance with their respective terms, all of Seller's duties and obligations arising on or after the beginning of business on the Closing Date with respect to the Acquired Loans purchased by Buyer (collectively, the "<u>Assumed Obligations</u>"), including, without limitation:

1. all obligations and duties with respect to servicing all Acquired Loans in the Loan Portfolio purchased hereunder, including the obligation to service such Acquired Loans in accordance with applicable law and commercially reasonable standards;

2. all of Seller's duties and obligations under the Acquired Loans as specified in the related Loan Documents and including the obligation to comply with and discharge all legal requirements and all obligations under the applicable Loan Documents, including compliance with legal requirements relating to unfair collection practices in connection with any of the Acquired Loans purchased by Buyer hereunder; and

3. all liabilities in respect of Taxes with respect to the Loan Portfolio other than (x) those Taxes specifically allocated to Seller pursuant to this Agreement and (y) Taxes for which Seller Parties are required to indemnify Buyer Indemnified Parties pursuant to the last sentence of Section 8.1.1 of this Agreement.

(b) <u>Excluded Obligations</u>. Seller acknowledges and agrees that it is responsible for all expenses related to the Loan Portfolio and activity thereon prior to the Closing Date. Notwithstanding anything to the contrary set forth in this Agreement, Seller shall retain, and shall be responsible for paying, performing and discharging when due all, and Buyer shall not assume or have any responsibility for any, liabilities of Seller other than the Assumed Obligations (the "<u>Excluded Obligations</u>").

2.1.4 <u>Schedule of Acquired Loans</u>. Subject to the following sentence of this Section 2.1.4, Buyer and Seller agree that Schedule A sets forth all of the Acquired Loans in the Loan Portfolio to be purchased by Buyer hereunder as of the date of this Agreement. Buyer and Seller agree that Seller shall, not later than three (3) days prior to the Closing Date, revise Schedule A hereto to reflect any loans to be (a) added to the Loan Portfolio as Acquired Loans following the origination of such loans after the date hereof or (b) removed from the Loan Portfolio pursuant to Section 7.1 of this Agreement or following the repayment of the entire unpaid principal amounts of such loans after the date hereof.

2.1.5 <u>Obligations of the Parties</u>. Notwithstanding the foregoing acknowledgements in this Section 2.1, as between the Parties, each Party's obligations with respect to any other Party shall be governed by this Agreement, including, without limitation, the right to indemnification which shall be governed by Article 8 hereof.

2.2 <u>Closing</u>. The Closing of the purchase and sale of the Loan Portfolio shall take place on the Closing Date at the offices of Paul, Hastings, Janofsky & Walker LLP, 75 East

55th Street, New York, NY 10022, at 12:00 P.M. New York City time, unless otherwise agreed by both Buyer and Seller.

2.3 <u>Purchase Price; Payment at Closing; Deposit; Post-Closing Adjustments to</u> Purchase Price; Adjustments and Allocations with respect to Payments on Acquired Loans.

2.3.1 <u>Purchase Price</u>. Subject to this Section 2.3, the consideration to be paid by Buyer to Seller for the sale of the Loan Portfolio shall consist of (x) the assumption by Buyer of the Assumed Obligations and (y) an aggregate amount in cash equal to:

(a) one hundred two percent (102%) of the unpaid principal balance of each Acquired Loan set forth on Schedule A as of the Closing Date, except for those loans set forth on Schedule A under the heading "Florida Loans";

(b) one hundred two and a half percent (102.5%) of the unpaid principal balance of each Acquired Loan set forth on Schedule A under the heading "Florida Loans" as of the Closing Date;

(c) *plus* one hundred and one half percent (100.5%) of the unpaid principal balance of each Acquired Loan set forth on Schedule A-1 as of the Closing Date;

(d) *plus* ninety nine and one half percent (99.5%) of the unpaid principal balance of each Acquired Loan set forth on Schedule A-2 as of the Closing Date;

(e) *plus* accrued but unpaid interest for each Acquired Loan;

(f) *less* an amount equal to a loan loss reserve of 0.50% of the unpaid principal balance of each Acquired Loan as of the Closing Date, except for those loans set forth on Schedule A under the heading "Florida Loans";

(g) *less* an amount equal to a loan loss reserve of 1.00% of the unpaid principal balance of each Acquired Loan as of the Closing Date set forth on Schedule A under the heading "Florida Loans";

(h) *less* the Deposit

(collectively, the "<u>Purchase Price</u>").

2.3.2 Payment at Closing.

(a) <u>Notification</u>. No later than three (3) days prior to the Closing Date, Seller shall notify Buyer in writing of Seller's estimate of the amount of the Purchase Price to be paid to Seller by Buyer on the Closing Date, including the amount of the Purchase Price allocated to each Acquired Loan, any adjustments to be made to Schedule A pursuant to Section 2.1.4 of this Agreement and the adjustments to the Purchase Price to be made pursuant to Section 2.3.5 hereof (the "<u>Preliminary Closing Date Statement</u>"). All of the Seller's calculations and amounts set forth in the Preliminary Closing Date Statement shall be subject to the review and reasonable agreement of Buyer.

(b) <u>Payment</u>. Subject to the terms and conditions of this Agreement, on the Closing Date, Buyer Parties shall remit payment of the agreed upon Purchase Price as estimated and set forth in the Preliminary Closing Date Statement (the "<u>Closing Date Payment Amount</u>") to Seller by wire transfer in accordance with the instructions provided by Seller to Buyer. The Closing Date Payment Amount may be adjusted after the Closing pursuant to Section 2.3.4(a) hereof.

2.3.3 <u>Deposit</u>. Upon the execution of this Agreement by Buyer and Seller, Buyer Parties shall deposit in escrow with the Escrow Agent pursuant to the provisions of the Escrow Agreement, a cash deposit in an amount equal to One Hundred Thousand Dollars (\$100,000) (the "<u>Deposit</u>"). The Deposit shall be held in an interest-bearing account approved by Seller and Buyer, and all interest thereon shall be deemed part of the Deposit. At Closing, the Deposit shall be credited against the Purchase Price.

2.3.4 <u>Post Closing Adjustments to the Purchase Price and Loan Portfolio</u>.

Post Closing Adjustments to the Purchase Price. Promptly as (a) practicable following the Closing Date, but in no event later than 30 days, Buyer shall prepare in good faith and deliver to Seller a final calculation of the Purchase Price and any adjustments thereto as of the Closing Date (the "Final Closing Date Statement"). Within thirty (30) days after the receipt of the Final Closing Date Statement (the "Objection Period"), the Seller shall (A) review the Final Closing Date Statement together with the workpapers used in the preparation thereof and (B) object to any item or items shown thereon. Until a Final Closing Date Statement is settled pursuant to this Section 2.3.4(a), Buyer shall (A) provide Seller and Seller's authorized representatives reasonable access during normal business hours to all relevant workpapers to the extent required to complete their review of the Final Closing Date Statement, and (B) cooperate with Seller and Seller's authorized representatives in connection with their reasonable requests regarding the review of the Final Closing Date Statement, including by providing on a timely basis all information necessary in reviewing the Final Closing Date Statement. Seller shall notify Buyer of any objections to the Final Closing Date Statement, setting forth a description of such objection and the dollar amount of such objection. If Seller does not object during the Objection Period, the Final Closing Date Statement shall be conclusive and binding on the Parties. If Seller objects during the Objection Period and Buyer and Seller are unable to resolve such objections within 20 days after delivery by Seller of its objections, then all disagreements shall be submitted for resolution to a certified public accounting firm of national standing who has not had a substantial relationship with either Buyer (or any of its Affiliates) or Seller (or any of its Affiliates) in the last two (2) years, and who is reasonably acceptable to Seller and Buyer (the The Independent Auditor shall be selected as promptly as "Independent Auditor"). practicable, but in no event later than 10 days following the expiration of such 20-day period. If Seller and Buyer cannot agree on the selection of the Independent Auditor within the 10day period, then Seller and Buyer shall request the American Arbitration Association, New York office, to appoint the Independent Auditor. The American Arbitration Association's appointment of the Independent Auditor shall be final and binding. Each Party agrees to execute, if requested by the Independent Auditor, a reasonable engagement letter, including customary indemnities. The Independent Auditor shall act as an arbitrator to determine, based solely on the provisions of this Section 2.3.4(a) and not more than two rounds of presentations by Seller and Buyer, and not by independent review, only those issues still in dispute and in making such determination shall select either Buyer's or Seller's determination of the disputed dollar amount with respect to each issue disputed. The Independent Auditor shall have up to 30 days after its appointment to resolve any disputes that are submitted to such Independent Auditor. The Final Closing Date Statement, either as agreed to by Buyer and Seller or as adjusted by the Independent Auditor pursuant to the preceding sentence, shall be final and binding upon the Parties. The fees and expenses of the Independent Auditor shall be shared equally by Seller and Buyer. If the Purchase Price as reflected on the Final Closing Date Statement ("Final Purchase Price") is less than the Purchase Price as estimated in the Preliminary Closing Date Statement and paid on the Closing Date, Seller shall pay to Buyer, by wire transfer in immediately available funds, the amount by which the Closing Date Payment Amount exceeds the Final Purchase Price, plus interest at an annual rate equal to 2.00%, calculated on a daily basis from and including the Closing Date to but excluding the date of payment. If the Final Purchase Price is greater than the Closing Date Payment Amount, Buyer shall pay to Seller, by wire transfer in immediately available funds, the amount by which the Final Purchase Price exceeds the Closing Date Payment Amount, plus interest at an annual rate equal to 2.00%, calculated on a daily basis from and including the Closing Date to but excluding the date of payment.

Post Closing Adjustments to the Loan Portfolio. Within six (6) (b) months after the Closing Date, but in no event after such time, Buyer shall prepare in good faith and deliver to Seller a written list of any loans acquired on the Closing Date that Buyer believes should have been excluded from the Loan Portfolio as Excluded Loans under Section 2.1.2(a)-(q) of this Agreement, together with a reference to the specific provisions of Section 2.1.2(a)-(q) under which Buyer believes each such loans should have been excluded from the Loan Portfolio (such list and reference, the "Adjustment Schedule"). If Seller disagrees with the Adjustment Schedule presented by Buyer hereunder or any aspect thereof, Seller may within fifteen (15) days after receipt of the Adjustment Schedule (the "Repurchase Objection Period") deliver a written notice to Buyer of such disagreement and specifying in reasonable detail Seller's basis for disagreement (the "Repurchase Objection Statement"). If Seller does not object during the Repurchase Objection Period, the Adjustment Schedule shall be conclusive and binding on the Parties. If Seller objects during the Repurchase Objection Period and Buyer and Seller are unable to resolve such objections within 20 days after delivery by Seller of the Repurchase Objection Statement, then all disagreements shall be submitted for resolution to the Independent Auditor, and resolved by such Independent Auditor, pursuant to the procedures set forth in Section 2.3.4(a) above. Following a final determination of the Adjustment Schedule, Seller shall be allowed an opportunity for a period of 30 (thirty) days to cure any non-conformity with respect to each subject loan, and if it is impossible to cure such non-conformity or Seller chooses not to cure or correct such nonconformity, Seller will promptly repurchase each such loan from Buyer for the Allocated Purchase Price, as adjusted based on any principal payments received, for each such loan; provided, however, that Seller shall not be required to repurchase loans for which the Allocated Purchase Price, in the aggregate, would exceed \$2,000,000.

2.3.5 Adjustments and Allocations With Respect to Payments on Acquired

Loans.

(a) All payments received by Seller on account of any of the Acquired Loans prior to the Closing Date shall belong to Seller and shall be reflected in Seller's calculation of the Preliminary Closing Date Statement. All payments received by Seller on account of any of the Acquired Loans on or after the Closing Date shall belong to Buyer, shall be held in trust for the benefit of Buyer and shall be paid to Buyer within a reasonable time after receipt, and in no event later than five (5) Business Days following receipt by Seller, with checks and other negotiable instruments being endorsed by Seller without recourse and without warranties. Buyer and Seller agree that Seller retains the right to collect on any and all checks, or other negotiable instruments, which represent payment of principal, interest, or any other fees or charges under any of the Acquired Loans, to the extent received by Seller prior to the Closing Date, credited by Seller to the applicable Obligor and not otherwise endorsed to Buyer as set forth above.

(b) In the event of the removal of any Acquired Loan prior to Closing pursuant to Section 7.1 hereof, the Purchase Price shall be adjusted by subtracting the Allocated Purchase Price for such removed Acquired Loan, and all allocations and adjustments under this Section 2.3 described above shall be made without inclusion of such removed Acquired Loan or any payments relating thereto.

2.4 Assignments and Endorsements.

2.4.1 Upon payment of the Purchase Price for the Loan Portfolio on the Closing Date in accordance with the terms of Section 2.3, Seller shall (i) execute and deliver to Buyer a bill of sale, in the form of Exhibit A hereto, and (ii) deliver any notes endorsed as set forth in Section 2.4.4 below.

2.4.2 Buyer shall prepare and deliver at Closing any further assignments or other documents (including any specific assignment agreements required in connection with any Acquired Loan where Seller is a participant) as Buyer or Seller deem necessary or appropriate pursuant to this Agreement for the legal transfer to Buyer of Seller's right, title and interest in and to the Loan Portfolio, all of which must be in form and substance reasonably acceptable to Seller. Buyer will accept a UCC-3 Assignment from Israel Discount Bank of New York, as Custodian under that certain agreement dated September 9, 1993 with Seller's banks and the SBA for each Acquired Loan as proof of the assignment to Buyer of a first priority security interest in each Acquired Loan (except for those loans set forth on Schedule 3.1.8). Buyer shall be responsible for the preparation of, and payment of any costs associated with the preparation of, such assignments and other documents, and for the recording of such assignments and for payment of any costs and recording fees associated with recording such assignments or other documents, other than Transfer Taxes (which shall be paid equally by the Parties in accordance with Section 5.4.2 of this Agreement). Other than as set forth in this Agreement, the Loan Portfolio to be sold to Buyer hereunder shall be sold without recourse, representation or warranty of any kind or any responsibility or liability whatsoever, and all such assignments and other transfer documentation shall contain a provision to that effect. To the extent that there is a conflict between the terms of any assignment or other agreement, instrument or documents delivered pursuant to this Article 2 and the terms of this Section 2.4.2 or any other provision of this Agreement, the terms of this

Section 2.4.2 and the other provisions of this Agreement shall control and prevail among the Parties.

2.4.3 Should any assignments in addition to those delivered pursuant to Sections 2.4.1 and 2.4.2 above be required by applicable law, Buyer shall prepare and submit such additional assignments to Seller for execution within six (6) months after the Closing Date. Buyer shall be responsible for the preparation of, and any costs associated with the preparation of, such additional assignments, as well as for any costs, transfer Taxes or recording or filing fees associated with the recording of such additional assignments. Any such assignments shall be without recourse, representation or warranty of any kind or any responsibility or liability whatsoever, and in a form acceptable to Seller's counsel.

2.4.4 Seller shall endorse any note evidencing an Acquired Loan purchased hereunder as follows:

Pay to the order of _____,

THE PAYEE AND THE UNDERSIGNED ARE PARTIES TO A CERTAIN LOAN PORTFOLIO SALE AND PURCHASE AGREEMENT DATED AS OF JULY 16, 2008. OTHER THAN AS SET FORTH THEREIN, THIS ENDORSEMENT IS MADE WITHOUT RECOURSE, REPRESENTATION OR WARRANTY OF ANY KIND OR ANY RESPONSIBILITY OR LIABILITY WHATSOEVER.

ELK ASSOCIATES FUNDING CORPORATION

By:		
Date:		

If this form of endorsement does not conform to the requirements of applicable federal, state or local law, the Parties agree to modify the form of endorsement to conform to such requirements; <u>provided</u> that, in any event, all such endorsements shall confirm that the transfer related thereto is without recourse, representation or warranty of any kind or any responsibility or liability whatsoever other than as set forth above. In the event that Seller does not have, or is unable to locate, the original promissory note(s), if any, with respect to any loan, Seller shall deliver to Buyer on the Closing Date, for each such missing original promissory note, either (a) the lost note affidavit or equivalent received by Seller from the transferor with respect thereto, or (b) a separate lost note affidavit prepared and executed by Seller (in such form as Seller may reasonably deem appropriate, including a statement to the effect that that Seller has been unable to locate, after the exercise of reasonable diligence, the applicable original promissory note, and without any further representation or recourse to Seller), and, in each case, Seller shall have no further obligation or liability with respect to such missing promissory note; <u>provided</u>, <u>however</u>, that the principal amount of such missing original promissory notes (including promissory notes for loans originated between the date hereof and the Closing Date) for which Seller delivers to Buyer the items described in (a) and (b) above shall not, in the aggregate, exceed \$500,000.

2.4.5 Notwithstanding anything to the contrary contained herein, Buyer agrees that Seller shall be entitled to retain one (1) copy of the Loan Files for its records, which shall, following Closing, be considered confidential information of Buyer and subject to the terms and conditions of the Confidentiality Agreement. Following the Closing, if Seller is involved in any litigation or regulatory inquiry concerning an Acquired Loan, Buyer will grant Seller reasonable access to the applicable loan file, with respect to information that occurred prior to the Closing Date.

2.5 <u>Survival of Responsibilities</u>. The responsibilities and agreements contained in this Article 2 shall survive the Closing.

2.6 <u>Reimbursement for Use of Seller's Employees</u>. With respect to any litigation or proceedings relating to any Acquired Loans in which Seller or its employees are requested by Buyer or required by subpoena, court order or otherwise, to testify or to prepare responses to subpoenas or other legal process, pleadings or discovery to which Buyer is a party, Seller shall be reimbursed by Buyer for all reasonable associated travel, lodging and per diem costs and expenses. Seller shall, in its sole and absolute discretion, determine and assign the personnel necessary to perform such acts. Buyer shall also reimburse Seller for all reasonable attorneys' fees incurred by Seller in connection with such assigned litigation or proceedings and with any litigation or proceedings relating to the Acquired Loans arising after the Closing Date.

True Sale; Security Agreement. The Parties intend that each conveyance 2.7 hereunder be a sale of the Acquired Loans from Seller to Buyer, free and clear of all security interest, liens (other than liens not yet due or payable) or other encumbrances or adverse claims, for accounting, regulatory tax and other purposes and not a financing secured by such property; and the beneficial interest in and title to the Acquired Loans shall not be a part of Seller's estate in the event of the filing of a bankruptcy petition by or against Seller under any bankruptcy law. However, if, notwithstanding the intent of Seller and Buyer, under the Uniform Commercial Code, any bankruptcy law or other applicable law, this transaction is deemed to be a financing arrangement, or it is otherwise determined that any conveyance hereunder is for any reason not considered a sale and that the beneficial interest in and title to the Acquired Loans and such Acquired Loans remain part of Seller's estate, the Parties intend that with respect to any such Acquired Loans, this Agreement shall constitute a security agreement (as defined in the applicable Uniform Commercial Code) under the Uniform Commercial Code, and Seller hereby grants to Buyer on the terms and conditions of this Agreement a first priority perfected security interest in and against all of Seller's right, title and interest in and to the Acquired Loans, and other property conveyed hereunder and all proceeds of any of the foregoing, now existing and hereafter acquired.

ARTICLE 3

REPRESENTATIONS AND WARRANTIES OF SELLER PARTIES

3.1 <u>Seller Parties' Representations and Warranties</u>. Each Seller Party, jointly and severally, hereby represents and warrants to Buyer Parties as follows:

3.1.1 <u>Organization; Good Standing</u>. Seller is a corporation licensed as a Small Business Investment Company by the SBA and is duly organized, validly existing and in good standing under the laws of the State of New York. Ameritrans is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.

3.1.2 <u>Authorization</u>. Seller Parties have all requisite power and authority to execute and deliver, and to perform their respective obligations under this Agreement, and this Agreement has been duly authorized by all necessary action on the part of Seller Parties. This Agreement has been duly executed and delivered by Seller Parties and is the valid and binding obligation of Seller Parties, enforceable against such party in accordance with its terms, except as may be limited by any bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or similar law affecting creditors' rights and remedies generally and, with respect to the enforceability thereof, by general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity).

3.1.3 <u>No Conflicts; Consents</u>. Neither the execution and delivery by Seller Parties of this Agreement nor compliance by Seller Parties with the terms and provisions hereof will violate, conflict with or result in a breach of any of the terms, conditions or provisions of (a) the certificate of incorporation, bylaws or any other equivalent organizational document of Seller or Ameritrans; (b)(i) any judgment, order, injunction, decree or ruling of any court or of any Governmental Entity or any law, statute or regulation, to which Seller or Ameritrans is subject, or (ii) any agreement, contract or commitment to which Seller or Ameritrans or their respective properties are subject, other than in the case of subsection (b), approvals of regulatory authorities which have been obtained or will be obtained prior to or on the Closing Date. Seller Parties are not subject to any agreement with any regulatory authority which would prevent the consummation by Seller Parties of the transactions contemplated by this Agreement or any related agreements.

3.1.4 <u>Title to Acquired Loans</u>. Except for those Encumbrances identified on Schedule 3.1.4 hereto that will be satisfied and released at or prior to Closing, Seller has good title to the Seller's interest in the Acquired Loans, free and clear of any Encumbrances.

3.1.5 <u>Modifications to Loan Documents</u>. Except as reflected in the Loan Files, Seller has not materially modified the Loan Documents relating to any Acquired Loan or executed any instrument of release, cancellation or satisfaction related thereto.

3.1.6 <u>Loan Files and Information Provided as to Loan Balance</u>. The Loan Portfolio includes the originals or true and correct copies of all material Loan Documents. Seller further represents that all materials prepared by Seller and including estimates of the

outstanding principal balance of each Acquired Loan and/or a description of the future funding obligations thereunder are true and correct in all material respects as of the Closing Date and accurately reflect in all material respects the terms and conditions of the Loan Documents.

3.1.7 <u>Additional Representations Regarding Acquired Loans and Loan</u> <u>Documents</u>.

(a) With respect to each Medallion Loan, Seller hereby represents and warrants to Buyer as follows:

1. None of the Medallion Loans constitute Excluded Loans pursuant to Section 2.1.2(a) through (q).

2. Such Medallion Loan and the Loan Documents pertaining to such Medallion Loan have not been extended, waived, amended or modified by Seller except in accordance with Seller's customary policies and procedures.

3. Such Medallion Loan has been underwritten in accordance with Seller's customary lending policies and procedures.

4. There exist Loan Documents with respect to such Medallion Loan that consist of the following documents:

(1) Promissory note (originally executed or, to the extent the original promissory note cannot be located by Seller, a lost note affidavit or equivalent as described in Section 2.4.4 of this Agreement);

(2) Security agreement;

(3) Original filed UCC-1 or certified filed UCC-1, or

other proof of UCC filing;

- (4) Guaranty, if applicable; and
- (5) Evidence of ownership from the applicable

regulatory authority;

5. There exists a Loan File with respect to such Medallion Loan that may contain the following documents:

- (1) Participation agreement, if applicable;
- (2) Modification agreement, if applicable;
- (3) UCC-3, if applicable;
- (4) Disbursement letter;

- (5) Credit application;
- (6) Other financial information that meets applicable

SBA criteria;

(7) Customer identification (passport, driver's

license), or such other identification that was sufficient for Seller to have made such Medallion Loan;

- (8) Lien searches;
- (9) Proof of incorporation, if applicable
- (10) Credit report; and
- (11) SBA forms.
- 6. Such Medallion Loan has not matured.

7. The Loan Documents for such Medallion Loan were prepared using Seller's customary loan documentation for the market in which such Medallion Loan was originated as of the time at which such Medallion Loan was originated.

8. As of the date of this Agreement, there are no proceedings or investigations pending or, threatened, before any court, regulatory body, administrative agency, or other tribunal or governmental instrumentality of which Seller has Knowledge, or to which Seller is a party (1) asserting the invalidity of the Loan Documents for such Medallion Loan, (2) seeking to prevent payment and performance of the Loan Documents for such Medallion Loan, or (3) seeking any determination or ruling that might materially and adversely affect the validity, enforceability or collectability of such Medallion Loan or the related Loan Documents.

9. As of the date of this Agreement, there are no loans to a single Obligor or related loans guaranteed by a common guarantor in excess of \$2.5 million.

10. Except as set forth on <u>Schedule 3.1.7(a)</u>, for non-balloon Medallion Loans, such Medallion Loan's final scheduled payment is due no later than 120 months after the date of its origination.

11. Except as set forth on <u>Schedule 3.1.7(a)</u>, for balloon Medallion Loans, such Medallion Loan's final scheduled payment is due no later than 120 months after the date of its origination.

12. Seller is the senior participant for such Medallion Loan to the extent such Medallion Loan is a participated Medallion Loan.

13. Such Medallion Loan complies with SBA regulations for loans made by a Small Business Investment Company.

14. Such Medallion Loan is not subject to any comment by a medallion regulatory agency or any other Governmental Entity.

(b) Any and all requirements of any federal, state or local laws applicable to Seller with respect to each Acquired Loan have been complied with.

(c) The Loan Documents have not been assigned or pledged, or, if previously assigned or pledged, have been released, or will be released at or prior to Closing with respect to those Encumbrances identified on Schedule 3.1.4 hereto, from any such pledge and reassigned to Buyer as described in Section 2.4.2 hereof.

(d) Except as disclosed to Buyer and accepted by Buyer in writing, there is no default, breach, violation or event of acceleration existing under the Loan Documents and to the best of Seller's Knowledge, no event which, with the passage of time or the giving of notice, or both, would constitute a default, breach, violation or event of acceleration; and Seller has not waived any default, breach, violation or event of acceleration.

(e) There is no pending or, to the Knowledge of Seller, threatened litigation with respect to any Acquired Loan which would adversely affect the rights of Buyer to enforce such Acquired Loan or otherwise obtain the benefits contemplated with respect to each Acquired Loan purchased.

(f) Seller has fulfilled and performed its obligations under each of the Loan Documents in all material respects in accordance with customary industry practice, Seller is not required to disburse any additional funds to any Obligor, except as specifically disclosed on Schedule 3.1.7(f) hereto, and Seller is not, and has received no written notice alleging that it is, in breach or default under, nor is there or is there alleged to be any basis for termination by any other party thereto of, any of the Loan Documents.

3.1.8 <u>Perfection of Collateral</u>. With respect to the Collateral, Seller has a perfected and valid first lien and first priority security interest in each item of Collateral as to which a security interest was granted under the collateral security agreements and other Loan Documents, unless otherwise disclosed on Schedule 3.1.8 hereto.

3.1.9 <u>Brokers' or Finders' Fees</u>. No Person acting on behalf of Seller Parties is or will be entitled to any brokers' or finders' fee or any other commission or similar fee, directly or indirectly, for which Buyer is liable or obligated in connection with any of the transactions contemplated hereby.

3.1.10 <u>Required Stockholder Vote</u>. The affirmative vote of (a) 66-2/3% of the shares of Seller's common stock outstanding on the record date for the Seller Stockholders Meeting and entitled to vote and (b) a majority of the shares of Ameritrans' common stock outstanding on the record date for the Ameritrans Stockholders Meeting and entitled to vote (together, the "<u>Required Stockholder Vote</u>") is the only vote of the holders of any class or series of Seller's capital stock and Ameritrans' capital stock necessary to adopt this Agreement and approve the transactions contemplated hereby.

3.1.11 <u>Litigation</u>. There are no actions, suits, investigations, or proceedings by or before any Governmental Entity pending, or, to the Knowledge of Seller or Ameritrans, threatened against Seller or Ameritrans which in any manner challenges the validity of this Agreement or any action taken by Seller or Ameritrans pursuant to this Agreement or seeks to prevent, enjoin, alter or materially delay any transaction contemplated hereby.

3.1.12 <u>No Impairment or Delay</u>. To the Knowledge of Seller Parties, neither Seller Party is subject to any condition or event that would reasonably be expected to (i) impair in any material respect the ability of Seller Parties to perform their respective obligations under this Agreement or (ii) prevent or materially delay the consummation of any of the transactions contemplated by this Agreement.

3.2 <u>Disclaimers</u>.

DISCLAIMER OF WARRANTIES AND REPRESENTATIONS NOT 3.2.1 EXPRESSED HEREIN. EXCEPT FOR THOSE REPRESENTATIONS AND WARRANTIES CONTAINED IN SECTION 3.1, NO WARRANTIES OR REPRESENTATIONS, EXPRESS OR IMPLIED, HAVE BEEN MADE BY SELLER OR BY ANYONE ACTING ON ITS BEHALF.

3.2.2 <u>Disclaimer of Accuracy/Completeness</u>. Neither Seller nor anyone acting on Seller's behalf has or will have any duty, either initially or on a continuing basis, to make any investigation, evaluation, appraisal of, or any responsibility or liability with respect to the accuracy or completeness of, any information provided to Buyers which has been provided to Seller or anyone acting on Seller's behalf by any Obligor. Nothing in this Agreement or in the Loan Documents shall impose upon Seller or anyone acting on Seller's behalf any fiduciary relationship in respect of Buyer.

ARTICLE 4

REPRESENTATIONS AND WARRANTIES OF BUYER PARTIES

4.1 <u>Buyer Parties' Representations and Warranties</u>. Buyer Parties, jointly and severally, hereby represent and warrant to Seller Parties as follows:

4.1.1 <u>Organization</u>; <u>Good Standing</u>. Buyer is an industrial bank duly organized, validly existing and in good standing under the laws of the State of Utah. MFC is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.

4.1.2 <u>Authorization</u>. Buyer Parties have all requisite power and authority to execute and deliver, and to perform Buyer Parties' obligations under, this Agreement, and this Agreement has been duly authorized by all necessary action on the part of Buyer Parties. This Agreement has been duly executed and delivered by Buyer Parties and is the valid and binding obligation of Buyer Parties, enforceable against Buyer Parties in accordance with its terms, except as may be limited by any bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or similar law affecting creditors' rights and remedies generally and, with respect to the enforceability thereof, by general principles of equity, including

principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity).

4.1.3 <u>No Conflicts; Consents</u>. Neither the execution and delivery by Buyer Parties of this Agreement nor compliance by Buyer Parties with the terms and provisions hereof will conflict with or result in a breach of any of the terms, conditions or provisions of (a) the certificate of incorporation or bylaws of Buyer Parties or (b)(i) any judgment, order, injunction, decree or ruling of any court or of any Governmental Entity or any law, statute or regulation, to which Buyer Parties are subject, or (ii) any agreement, contract or commitment to which Buyer Parties or their properties are subject, the breach of which judgment, order, injunction, decree, ruling, law, statute, regulation, agreement, contract or commitment would have an adverse effect on the ability of Buyer Parties to execute, deliver or perform their obligations pursuant to this Agreement.

4.1.4 <u>Financial Ability</u>. Buyer has, and at the Closing Date will have, the financial means and ability to purchase the Loan Portfolio and to assume and discharge all of the obligations of Seller pursuant to each and every Acquired Loan in the Loan Portfolio.

4.1.5 <u>Brokers' or Finders' Fees</u>. No Person acting on behalf of Buyer Parties are or will be entitled to any brokers' or finders' fee or any other commission or similar fee, directly or indirectly, for which Seller is liable or obligated in connection with any of the transactions contemplated hereby.

4.1.6 <u>Litigation</u>. There are no actions, suits, investigations, or proceedings by or before any Governmental Entity pending, or, to the Knowledge of Buyer Parties, threatened against Buyer Parties which in any manner challenges the validity of this Agreement or any action taken by Buyer Party pursuant to this Agreement or seeks to prevent, enjoin, alter or materially delay any transaction contemplated hereby.

4.1.7 <u>No Impairment or Delay</u>. To the Knowledge of Buyer Parties, neither Buyer Party is subject to any condition or event that would reasonably be expected to (i) impair in any material respect the ability of either Buyer Party to perform its obligations under this Agreement or (ii) prevent or materially delay the consummation of any of the transactions contemplated by this Agreement.

ARTICLE 5

COVENANTS OF SELLER PARTIES AND BUYER PARTIES

5.1 <u>General</u>. Each of Seller Parties and Buyer Parties shall use commercially reasonable efforts to take all action and to do all things necessary, proper, or advisable in order to consummate and make effective the transactions contemplated by this Agreement.

5.2 <u>Proxy Statement</u>.

5.2.1 As promptly as practicable after the date hereof, Seller Parties shall prepare, and Buyer Parties shall provide reasonable cooperation to Seller Parties in the preparation of, a proxy statement to be sent to Ameritrans's stockholders in connection with

the Ameritrans Stockholders Meeting (together with any amendment thereof or supplements thereof, the "Proxy Statement"). As promptly as practicable after the date hereof and from time to time thereafter, Buyer Parties shall provide such information regarding Buyer Parties and their Affiliates that Seller Parties may reasonably request for inclusion in the Proxy Statement. Subject to the timely receipt of any information regarding Buyer Parties that is required to be included in the Proxy Statement, Seller Parties shall: (a) cause the Proxy Statement to comply with the rules and regulations promulgated by the SEC; (b) provide Buyer Parties with a reasonable opportunity to review and comment on drafts of the Proxy Statement; (c) promptly cause the Proxy Statement to be filed with the SEC; (d) promptly provide Buyer Parties with copies of all correspondence between Seller Parties, Ameritrans or any of their representatives, on the one hand, and the SEC, on the other hand; (e) promptly notify Buyer Parties upon the receipt of any comments or requests from the SEC with respect to the Proxy Statement; (f) promptly respond to any comments or requests of the SEC; and (g) cause the Proxy Statement to be mailed to Ameritrans' stockholders as promptly as practicable following the date hereof and clearance of the Proxy Statement by the SEC.

5.2.2 If any event relating to Seller Parties occurs, or if Seller Parties become aware of any information, that should be disclosed in an amendment or supplement to the Proxy Statement, then Seller Parties shall promptly inform Buyer Parties of such event or information and shall, in accordance with the procedures set forth in Section 5.2.1, (a) prepare and file with the SEC such amendment or supplement as soon thereafter as is reasonably practicable, and (b) if appropriate, cause such amendment or supplement to be mailed to the stockholders of Ameritrans. If any event relating to Buyer Parties or their Affiliates occurs, or if Buyer becomes aware of any information regarding Buyer Parties or their Affiliates that is required to be disclosed in an amendment or supplement to the Proxy Statement, then Buyer Parties shall promptly inform Seller Parties of such information.

5.2.3 Seller Parties shall ensure that none of the information included or incorporated by reference in the Proxy Statement (other than information provided by Buyer Parties and included in the Proxy Statement) will, at the time the Proxy Statement is mailed to the stockholders of Ameritrans or at the time of the Ameritrans Stockholders Meeting (or any adjournment or postponement thereof), contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they are made, not misleading.

5.2.4 Buyer Parties shall ensure that none of the information provided by Buyer Parties and included in the Proxy Statement will, at the time the Proxy Statement is mailed to the Ameritrans' stockholders or at the time of the Ameritrans Stockholders Meeting (or any adjournment or postponement thereof), contain any untrue statement of a material fact or omit to state any material fact required to be included in such information or necessary in order to make statements therein, in light of the circumstances under which they are made, not misleading.

5.2.5 The proxy materials will be sent to the Ameritrans' stockholders for the purpose of soliciting proxies from Ameritrans' stockholders to vote in favor of (i) the adoption of this Agreement and the approval of the transactions contemplated herein and (ii)

any other matters that may be properly brought before the Ameritrans' stockholders to be voted thereon. Ameritrans, acting through its board of directors, shall include in the Proxy Statement the recommendation of its board of directors that the Ameritrans stockholders vote in favor of the adoption of this Agreement and the approval of the Transaction.

5.3 <u>Governmental and Third-Party Consents</u>. Seller Parties and Buyer Parties shall use commercially reasonable efforts to seek consents of any Governmental Entity or any other third party required to be obtained in order to consummate the transactions contemplated by this Agreement.

5.4 <u>Tax Matters</u>.

5.4.1 Allocation of Purchase Price. Seller and Buyer agree to allocate the Purchase Price (including any adjustments thereto) and Assumed Obligations for all purposes (including Tax and financial accounting purposes) in a manner consistent with Code Section 1060 and the Treasury Regulations thereunder. If Buyer and Seller agree on such allocation (the "Allocation"), within one hundred twenty (120) days after the Closing Date (which shall be evidenced by an allocation schedule signed by each of Buyer and Seller), Buyer and Seller shall (i) be bound by the Allocation for purposes of determining any Taxes, (ii) timely file all forms (including Internal Revenue Service Form 8594) and Tax returns required to be filed in connection with the Allocation, (iii) prepare and file, and cause its Affiliates to prepare and file, its Tax returns on a basis consistent with the Allocation, and (iv) take no position, and cause its Affiliates to take no position, inconsistent with the Allocation on any applicable Tax Return, in any proceeding before any taxing authority or in any report made for Tax purposes. In the event that the Allocation is disputed by any taxing authority, the Party receiving notice of the dispute shall promptly notify and consult with the other Parties and keep the other Parties reasonably appraised of material developments concerning resolution of such dispute. In the event that Buyer and Seller are unable to agree on such allocation within one hundred twenty (120) days after the Closing Date, the Independent Auditor selected pursuant to Section 2.3. of this Agreement shall, at Seller and Buyer's joint and equal expense, determine the appropriate allocations based solely on presentations of Buyer and Seller (and not by independent review) within thirty (30) days after the expiration of the one hundred twenty (120) day period provided in the previous sentence.

5.4.2 <u>Taxes Resulting from Transaction</u>. Seller and Buyer agree to each pay on a timely basis fifty percent (50%) of all applicable transfer, sales, use, recording, registration and other similar Taxes as levied by any taxing authority or Governmental Entity in connection with the transactions contemplated by this Agreement (other than Taxes measured by or with respect to net income imposed on Seller or its Affiliates) (collectively, the "Transfer Taxes"). Buyer shall prepare, execute and file all Tax Returns, questionnaires, applications or other documents with respect to any Transfer Taxes in connection with the transactions contemplated by this Agreement. Buyer shall provide Seller with a copy of such Tax Returns no later than ten (10) business days prior to the date of filing. Upon approval, Seller shall submit Seller's portion of the Transfer Taxes (calculated pursuant to this Section 5.4.2), along with an executed Tax Return with respect to the Transfer Taxes, as applicable, to Buyer within ten (10) business days of receipt. Buyer and Seller shall jointly participate in the defense and settlement of any audit of, dispute with taxing authorities regarding, and any judicial or administrative proceeding relating to the liability for Transfer Taxes incurred in connection with this Agreement; <u>provided</u>, that neither Party shall settle any such audit, examination or proceeding without the prior written consent of the other Party, which consent shall not be unreasonably withheld. Each Party shall bear its own costs in participating in any such audit, examination or proceeding.

5.5 <u>Confidentiality</u>. The Parties acknowledge that Buyer and Seller have previously executed a confidentiality agreement, dated as of March 4, 2008 (the "<u>Confidentiality Agreement</u>"), which Confidentiality Agreement shall continue in full force and effect in accordance with its terms.

5.6 <u>Non-Competition</u>.

5.6.1 <u>Non-Competition of Seller Parties</u>. Subject to 5.6.4 of this Agreement, for a period of three (3) years following the Closing Date, Seller Parties shall not, in any Restricted Market, directly or indirectly, either as a principal, partner, agent, manager, stockholder or in any other capacity, engage in any activities constituting a Restricted Activity; <u>provided</u>, <u>however</u>, that ownership of less than five percent (5%) of any publicly held entity engaged in the Restricted Activity shall not constitute a violation hereof. Notwithstanding the preceding sentence and for the sake of clarity, Seller Parties shall at all times be permitted to engage in any business which they presently or hereinafter may conduct other than the Restricted Activity within the Restricted Market.

5.6.2 <u>Non-Competition of Seller Parties with respect to Obligors</u>. Subject to Section 5.6.4 of this Agreement and notwithstanding Section 5.6.1 of this Agreement, for a period of five (5) years following the Closing Date, Seller Parties shall not, directly or indirectly, either as a principal, partner, agent, manager, stockholder or in any other capacity, engage in any activities constituting a Restricted Activity involving any Obligor, including any Obligor during the period commencing three (3) months prior to the Closing Date and ending on the Closing Date; <u>provided</u>, <u>however</u>, that it is understood that this subsection shall not prohibit Seller Parties from engaging in any activities other than a Restricted Activity that involves an Obligor.

5.6.3 <u>Non-Compete of Gary C. Granoff</u>. For a period commencing on the Closing Date and ending June 30, 2011, Gary C. Granoff shall not, in any Restricted Market, directly or indirectly, either as a principal, partner, agent, manager, stockholder or in any other capacity, engage in any activities constituting a Restricted Activity; <u>provided</u>, <u>however</u>, that ownership of less than five percent (5%) of any publicly held entity engaged in the Restricted Activity shall not constitute a violation hereof; <u>provided</u>, <u>further</u>, that the restrictions set forth in this Section 5.6.3 shall remain in full force and effect for the period described above notwithstanding (a) the termination of Gary C. Granoff's employment under the Amended and Restated Employment Agreement, dated as of September 20, 2007, between Gary C. Granoff, Seller and Ameritrans (the "<u>Granoff Employment Agreement</u>") for cause or as a result of Gary C. Granoff's voluntary resignation or (b) the termination of Gary C. Granoff's employment than for cause or woluntary resignation to the extent that Seller Parties pay to Gary C. Granoff all amounts to which he is entitled pursuant to the Granoff Employment Agreement as a result of such

termination of his employment; and <u>provided</u>, <u>further</u>, that the restrictions set forth in this Section 5.6.3 shall terminate and shall be of no further force and effect upon the termination of Gary C. Granoff's employment under the Granoff Employment Agreement for any reason other than for cause or voluntary resignation to the extent that Seller Parties fail to pay to Gary C. Granoff all amounts to which he is entitled pursuant to the Granoff Employment Agreement as a result of such termination of his employment (the "Trigger Event"), in which event Seller Parties agree to pay to Buyer as liquidated damages an amount equal to (x) \$1,500,000 multiplied by (y) a fraction, the numerator of which shall be the number of months remaining between July 1, 2008 and June 30, 2011, and the denominator of which shall be 36 (for example, on August 1, 2008, the fraction shall be 35/36). Notwithstanding anything in this Section 5.6.3 to the contrary and for the sake of clarity, Gary C. Granoff shall at all times be permitted to engage in any business which he presently or hereinafter may conduct other than the Restricted Activity within the Restricted Market.

5.6.4 Non-Competition Exclusions.

(a) Notwithstanding anything in this Agreement to the contrary, no restrictive covenant or non-competition covenant on the part of Seller Parties shall in any way restrict the law firm of Granoff, Walker & Forlenza PC, solely from representing or providing legal counsel to any client in the normal course of its law practice, even if such client is making loans in the Restricted Markets concerning Restricted Activities. This paragraph 5.5.3 shall apply to any attorney employed by said firm and to the principals of said firm, including Gary C Granoff, Ellen M. Walker, or Lee Forlenza in regard to their representation of clients solely in an attorney-client relationship.

(b) Notwithstanding anything to the contrary contained herein, the non-competition covenants set forth in this Section 5.6 shall not restrict any activities of Seller Parties with respect to (1) any loans not included in the Loan Portfolio, including loans withdrawn from the Loan Portfolio pursuant to Section 7.1 hereof, or (2) any loans repurchased by Seller pursuant to Section 2.3.4(b) hereof, in each case, that relate to the refinancing or servicing of such loans.

5.6.5 Seller Parties acknowledge that the restrictions and agreements contained in this Section 5.6 are reasonable and necessary to protect the legitimate interests of Buyer, and that any violation of this Section 5.6 may cause substantial and irreparable injury to Buyer that may not be quantifiable and for which no adequate remedy may exist at law and agrees that injunctive relief, in addition to all other remedies, shall be available therefor.

5.6.6 It is the intent and understanding of each Party that if, in any action before any court or agency legally empowered to enforce this Section 5.6, any term, restriction, covenant, or promise is found to be unreasonable and for that reason unenforceable, then such term, restriction, covenant, or promise shall not thereby be terminated but that it shall be deemed modified to the extent necessary to make it enforceable by such court or agency and, if it cannot be so modified, that it shall be deemed amended to delete therefrom such provision or portion adjudicated to be invalid or unenforceable.

5.7 <u>Conduct of Business Prior to Closing</u>. Seller agrees that between the date hereof and Closing, as applicable, it shall, except as may be otherwise permitted or required by this Agreement, manage the Loan Portfolio in the ordinary course of business and use commercially reasonable efforts consistent with good business practices to preserve the good will of the Obligors; provided, that, all loans underwritten by Seller between the date hereof and Closing shall bear an interest rate no less than 7.5%, net, except in Miami, which loans shall bear an interest rate of no less than 9%, net; provided, further, that all loans underwritten by Seller between the date hereof and Closing in Miami shall, in the case of (x) a refinance, provide for a loan to value ratio of no more than 70% or (y) a new purchase, provide for a loan to value ratio of no more than 75%, in each case assuming a collateral value of \$190,000.

5.8 <u>Expenses</u>. All costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall, except as otherwise expressly provided herein, be paid by the party incurring such expenses, whether or not such transactions are consummated. Buyer hereby agrees to file all necessary documents (including all Tax Returns) with respect to all such amounts in a timely manner.

5.9 Additional Seller's Covenants to Buyer

5.9.1 <u>Further Document Deliveries</u>. Within ten (10) Business Days of the receipt by Seller of any Loan Documents not within the control and possession of Seller as of the Closing Date, including original copies of promissory notes in respect of the Acquired Loans, Seller shall deliver such Loan Documents to Buyer. Each such promissory note in respect of the Acquired Loans delivered pursuant to this Section 5.9 shall be duly endorsed, in accordance with Section 2.4.4, by Seller.

5.9.2 Seller Stockholders Meeting and Ameritrans Stockholders Meeting. Seller shall take all action necessary under all applicable legal requirements to call, give notice of and hold a meeting of Seller's stockholders to vote on the approval of this Agreement and the transactions contemplated hereby (the "<u>Seller Stockholders Meeting</u>"). Ameritrans shall take all action necessary under all applicable legal requirements to call, give notice of and hold a meeting of Ameritrans's stockholders to vote on the approval of this Agreement and the transactions contemplated hereby (the "<u>Ameritrans Stockholders</u> <u>Meeting</u>"). The Seller Stockholders Meeting and the Ameritrans Stockholders Meeting shall be held as promptly as practicable following the date hereof. Ameritrans shall ensure that all proxies solicited in connection with the Ameritrans Stockholders Meeting are solicited in substantial compliance with all applicable legal requirements. Each of Seller and Ameritrans agrees to use its commercially reasonable efforts to take all other action necessary or advisable to secure the Required Stockholder Vote.

5.9.3 <u>Access and Investigation</u>. During the period commencing on the date hereof and ending on the earlier of the termination of this Agreement and the Closing Date, Seller shall, following a request from Buyer to Seller in accordance with this Section 5.9.3, provide Buyer and its representatives with reasonable access during normal business hours to the Loan Files at the offices of Seller; <u>provided</u>, that any information shared by Seller with Buyer and its representatives pursuant to this Section 5.9.3 shall be kept confidential, shall not be disclosed to any third parties except those of its representatives who are required to have such information for the purposes described in this Section 5.9.3, and shall be used solely by Buyer Parties for the purposes of (i) consummating the transactions contemplated by this Agreement and (ii) assuming all obligations and liabilities described herein under the Loan Portfolio following the Closing. Buyer shall schedule and coordinate all inspections only with Michael Feinsod and shall request such matters a reasonable amount of time in advance, specifying the inspection or materials that Buyer or Buyer's representatives intend to conduct. Seller shall be entitled to have its representatives present at all times during any such inspection and the inspections shall take place at mutually convenient times. Buyer agrees that it will not take any notes from the inspection room that would reveal the name, address, telephone number or any other proprietary information concerning the identity of the borrower or guarantor of the loans. Notwithstanding the foregoing, Seller shall not be required to provide access to or to disclose information where such access or disclosure would jeopardize the attorney-client privilege of Seller or contravene any legal requirement or binding agreement entered into prior to the date hereof. The relevant parties will make appropriate substitute disclosure arrangements under circumstances in which the restrictions of the preceding sentence apply.

5.10 Additional Buyer's Covenants to Seller.

5.10.1 <u>Notification of Obligors</u>. Seller shall cooperate with Buyer in good faith to enable Buyer to prepare, print and mail, at Buyer's expense, a notice notifying each Obligor on a timely basis of the purchase of the acquired Loan Portfolio by Buyer and such other information as may be required to be given to such Obligor and other matters which the Parties determine to be appropriate. Any such notice shall be in a form consented to by each of the Parties prior to mailing, but no Party shall unreasonably withhold or delay such consent.

5.10.2 <u>Use of Seller's Names</u>. Buyer shall not, without the express written consent of Seller, institute any legal action in the name of Seller. Buyer shall not, through misrepresentation or nondisclosure, mislead or conceal from any Person the identity of the owner of the Acquired Loans purchased under this Agreement. Buyer shall not use or refer to Seller, or any name derived from the name of Seller or confusingly similar therewith, to promote Buyer's sale, collection or management of the Acquired Loans purchased under this Agreement. Buyer acknowledges that there is no adequate remedy at law for violation of this Section 5.10.2 and in addition to all other equitable remedies, Seller shall be entitled to specific performance and injunctive relief and other equitable relief as a remedy for any such breach.

5.10.3 <u>Informational Tax Reporting</u>. Buyer agrees to assume, as of the Closing Date, all obligations with respect to federal, state and local income tax informational reporting related to the Acquired Loans. Seller agrees to prepare all Tax Returns with respect to the Acquired Loans that end on or prior to the Closing Date. Buyer further agrees to cooperate with Seller to the extent necessary to allow Seller to fulfill its obligations with respect to such Tax Return reporting for such Acquired Loans for all periods ending on or prior to the Closing Date.

5.10.4 <u>Notice of Claims</u>. Buyer shall promptly notify Seller of any claim, threatened claim or litigation against Seller (or any prior holder of the Loans) arising out of any Acquired Loan or any Assumed Obligations that may come to Buyer's attention.

ARTICLE 6

CONDITIONS TO CLOSING AND TERMINATION

6.1 <u>Conditions to Closing</u>.

6.1.1 <u>Conditions to the Obligations of Each Party</u>. The respective obligations of each Party to effect the purchase of the Loan Portfolio are subject to the conditions precedent that (i) no injunction, order, decree or ruling issued by a Governmental Entity, nor any statute, rule, regulation or executive order promulgated or enacted by any Governmental Entity, shall enjoin, make illegal or otherwise materially impair or restrict such assignment, assumption or consummation, as the case may be, (ii) all required consents of any Governmental Entity or any other third party required to be obtained in order to consummate the transactions contemplated by this Agreement shall have been obtained and be in full force and effect, and all required waiting periods and any extensions thereof, shall have expired or been terminated, (iii) no action, suit, investigation or proceeding by any Governmental Entity shall have been instituted or threatened to restrain or prohibit or otherwise challenge the legality or validity of the transactions contemplated by this Agreement, (iv) this Agreement shall have been duly approved by the Required Stockholder Vote and (vi) the Specified Diversified Loan Event shall have occurred.

6.1.2 <u>Additional Conditions to the Obligations of Buyer Parties</u>. The obligations of Buyer Parties are also subject to fulfillment (or written waiver by Buyer Parties) at or prior to the Closing Date of each of the following conditions precedent:

(a) <u>Representations and Warranties</u>. Subject to and after application of Section 7.1, the representations and warranties contained in Article 3 shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date as though made at and as of the Closing Date.

(b) <u>Performance of Covenants</u>. Each Seller Party shall have duly performed and complied in all material respects with each covenant and agreement required to be performed or complied with by it under this Agreement, in each case, as and when required to be performed or complied with by it prior to or on the Closing Date.

(c) <u>Certificate of Ameritrans</u>. Ameritrans shall have delivered to Buyer a certificate signed by an officer of Ameritrans to the effect that each of the conditions specified in Sections 6.1.2(a) and (b) of this Agreement has been satisfied in all respects.

(d) <u>Closing Documents</u>. Buyer shall have received each of the Closing Documents to be delivered by Seller Parties at or prior to the Closing.

(e) <u>Opinion</u>. Buyer shall have received a favorable opinion from counsel to the Seller in Seller counsel's standard form covering valid existence and good

standing, corporate power and authority, enforceability, non-contravention of organizational documents, no violation of law and qualification to do business in New York.

6.1.3 <u>Additional Conditions to the Obligations of Seller Parties</u>. The obligations of Seller Parties are also subject to fulfillment (or written waiver by Seller Parties) at or prior to the Closing Date of each of the following conditions precedent:

(a) <u>Representations and Warranties True</u>. The representations and warranties contained in Article 4 shall be true and correct in all material respects, in each case, as of the date of this Agreement and as of the Closing Date as though made at and as of the Closing Date.

(b) <u>Performance of Covenants</u>. Each Buyer Party shall have duly performed and complied in all material respects with each covenant and agreement required to be performed or complied with by it under this Agreement, in each case, as and when required to be performed or complied with by it prior to or on the Closing Date.

(c) <u>Certificate of MFC</u>. MFC shall have delivered to Seller a certificate signed by an officer of MFC to the effect that each of the conditions specified in Sections 6.1.3(a) and (b) of this Agreement has been satisfied in all respects.

(d) <u>Closing Documents</u>. Seller shall have received each of the Closing Documents to be delivered by Buyer Parties at or prior to the Closing.

6.2 <u>Termination</u>.

6.2.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Closing Date:

(a) by mutual agreement of Seller and Buyer;

(b) by Seller if a material breach of any provision of this Agreement has been committed by any Buyer Party and such breach has not been waived in writing by Seller or cured by such Buyer Party within ten (10) days after receipt of written notice from Seller requesting such breach to be cured;

(c) subject to Section 7.1, by Buyer if a material breach of any provision of this Agreement has been committed by any Seller Party and such breach has not been waived in writing by Buyer or cured by such Seller Party within ten (10) days after receipt of written notice from Buyer requesting such breach to be cured (or the applicable Acquired Loans have not been removed pursuant to Section 7.1);

(d) subject to Section 7.1, (i) by Buyer if any of the conditions of Section 6.1.1 and 6.1.2 have not been satisfied on or before October 15, 2008 or if satisfaction of such a condition is or becomes impossible (other than through the failure of Buyer to comply with its obligations under this Agreement) and Buyer has not waived such condition in writing on or before such time; or (ii) by Seller if any of the conditions of Sections 6.1.1 and 6.1.3 have not been satisfied on or before October 15, 2008 or if satisfaction of such a

condition is or becomes impossible (other than through the failure of Seller to comply with its obligations under this Agreement) and Seller has not waived such condition in writing on or before such time; or

(e) by Seller on the one hand or Buyer on the other hand, if any Governmental Entity shall have issued a final permanent order enjoining or otherwise prohibiting the transactions contemplated by this Agreement, and the time for appeal or petition for reconsideration of such order shall have expired.

6.2.2 <u>Notice of Termination</u>. Any party desiring to terminate this Agreement pursuant to this Section 6.2 shall give written notice of such termination to the other party to this Agreement.

6.2.3 <u>Effect of Termination</u>. If this Agreement is terminated, all further obligations of the Parties or their respective Related Persons shall be terminated without further liability of any party to the other (except as set forth in Sections 5.5, 5.8, this Section 6.2 and Article 9 hereof), and Seller shall have the right to market or sell the Loan Portfolio free of any rights by Buyer under this Agreement; <u>provided</u>, <u>however</u>, that nothing herein shall relieve any party from liability for its fraudulent or intentional misrepresentation, omission or breach of this Agreement or any related agreement or for payment of the amounts set forth in Section 6.2.4 of this Agreement, as applicable.

6.2.4 <u>Release of Deposit Upon Termination</u>. In the event of any termination by Seller of this Agreement pursuant to Section 6.2.1(b), Seller shall be entitled to (a) the full principal amount of the Deposit which shall be released by the Escrow Agent to Seller pursuant to the terms of the Escrow Agreement and (b) to pursue all legal and equitable remedies against Buyer Parties for any and all losses sustained or incurred as a result of such Buyer's breach of this Agreement. In such event, interest on the Deposit shall be released by the Escrow Agent to Seller. Buyer shall be entitled to a return of the Deposit and interest thereon if this Agreement is terminated for any other reason.

6.2.5 <u>Survival</u>. The obligations of the Parties to this Agreement under Sections 5.5, 5.8, this Section 6.2 and Article 9 shall survive any such termination.

ARTICLE 7

BUYER PRE-CLOSING REMEDY

7.1 <u>Pre-Closing Remedy</u>. If, prior to the Closing Date, any Buyer Party or Seller Party discovers the breach of any of the representations, warranties or covenants of Seller Parties set forth in Article 3 of this Agreement as to a particular Acquired Loan, or the conditions precedent to Buyer's obligations to purchase any particular Acquired Loan have not been satisfied, the party discovering such matter will promptly but in no event after the Closing Date give notice of such discovery to the other party, along with a description of the breach or applicable condition. Seller shall be allowed an opportunity to cure any breach or condition, and if it is impossible to cure such breach or Seller chooses not to cure or correct the breach or meet such condition, prior to the Closing Date, Seller shall withdraw the subject Acquired Loan from sale and the Purchase Price will be reduced as set forth in Section 2.3.5(b) hereof.

ARTICLE 8

INDEMNIFICATION

8.1 <u>Seller Parties' Indemnification Obligations</u>.

8.1.1 Subject to Sections 8.3 and 8.6.1 hereof, after the Closing, Seller Parties shall jointly and severally indemnify and hold harmless Buyer Parties and their respective Affiliates and their respective officers, directors and employees (each a "Buyer Indemnified Party" and together, the "Buyer Indemnified Parties"), from and against any and all Losses arising from or relating to: (i) any breach of any representation or warranty of Seller Parties contained in this Agreement or the Specified Diversified Loan Sale Agreement (without giving effect to any materiality qualifier contained herein or therein); (ii) any breach of any covenants of Seller Parties contained in this Agreement or the Specified Diversified Loan Sale Agreement; or (iii) any Excluded Obligation arising from this Agreement or the Specified Diversified Loan Sale Agreement. Except as otherwise provided in this Agreement, Seller Parties shall also indemnify and hold harmless the Buyer Indemnified Parties for all Taxes with respect to the Loan Portfolio in this Agreement or the Specified Diversified Loan Sale Agreement for any taxable period ending on or prior to the Closing. For the avoidance of doubt (and notwithstanding anything otherwise provided herein), the Parties acknowledge and agree that it is the intent of the Parties that Buyer shall assume all credit risk associated with the Acquired Loans. Accordingly, Seller shall have no indemnification obligation hereunder if Buyer is (a) unable to collect from an Obligor of an Acquired Loan or (b) otherwise unable to enforce the terms of an Acquired Loan against an Obligor thereunder; provided, in either case, that Buyer's inability did not result from Seller's breach of a covenant, representation or warranty made by Seller in this Agreement.

8.2 <u>Buyer Parties' Indemnification Obligations</u>.

8.2.1 Subject to Sections 8.3 and 8.6.2 hereof, after the Closing, Buyer Parties shall jointly and severally indemnify and hold harmless Seller Parties and their respective Affiliates and their respective officers, directors and employees (each a "<u>Seller Indemnified Party</u>" and together, the "<u>Seller Indemnified Parties</u>"), from and against any and all Losses arising from or relating to: (i) the breach of any representation or warranty of Buyer Parties contained in this Agreement or the Specified Diversified Loan Sale Agreement (without giving effect to any materiality qualifier contained herein or therein); (ii) any breach of any covenants of Buyer Parties contained in this Agreement or the Specified Diversified Diversified Loan Sale Agreement; or (iii) any Assumed Obligation arising from this Agreement or the Specified Diversified Loan Sale Agreement, Buyer Parties shall also indemnify and hold harmless the Seller Indemnified Parties for all Taxes assessed against Buyer Parties with respect to the Loan Portfolio in this Agreement or the Specified Diversified Loan Sale Agreement for any taxable period not included in the last sentence of Section 8.1.

8.3 <u>Obligation to Mitigate Losses</u>. Buyer Indemnified Parties and Seller Indemnified Parties shall use their commercially reasonable efforts to mitigate any actual or potential Loss, in each case to the same extent as it would if such Loss were not subject to indemnification pursuant to the terms of this Agreement.

8.4 <u>Tax Consequences of Indemnification</u>. Buyer and Seller agree that, unless otherwise required under applicable law, any indemnification payment and any other payment made pursuant to this Agreement shall be treated as an adjustment to the Purchase Price for all Tax purposes.

8.5 <u>Procedures</u>.

8.5.1 <u>Notice of Claims</u>. The Parties agree that in case any claim is made, any suit or action is commenced, or any Knowledge is received of a state of facts which, if not corrected, may give rise to a right of indemnification for an Indemnified Party from the other Party ("<u>Indemnifying Party</u>"), the Indemnified Party will give notice to the Indemnifying Party as promptly as practicable after the receipt by the Indemnified Party of notice or Knowledge of such claim, suit, action or state of facts. Notice to the Indemnifying Party under the preceding sentence shall be given no later than thirty (30) days after receipt by the Indemnified Party of service of process in the event a suit or action has commenced or sixty (60) days under all other circumstances. The failure to give prompt notice shall not relieve an Indemnifying Party of its obligation to indemnify except to the extent the Indemnifying Party is prejudiced by such failure.

8.5.2 <u>Inspection and Assistance</u>. The Indemnified Party shall make available to the Indemnifying Party and its counsel and accountants at reasonable times and for reasonable periods, during normal business hours, all books and records of the Indemnified Party relating to any such possible claim for indemnification, and each Party will render to the other Party such assistance as it may reasonably require of the other in order to insure prompt and adequate defense of any suit, claim or proceeding based upon a state of facts which may give rise to a right of indemnification hereunder.

8.5.3 <u>Right to Defend</u>. The Indemnifying Party shall have the right to defend, compromise and settle any third party suit, claim or proceeding in the name of the Indemnified Party to the extent that the Indemnifying Party may be liable to the Indemnified Party in connection therewith. The Indemnifying Party shall notify the Indemnified Party within thirty (30) days of having been notified pursuant to this Section 8.5 if the Indemnifying Party elects to assume the defense of any such claim, suit or proceeding and employ counsel. The Indemnified Party shall have the right to employ its own counsel if the Indemnifying Party so elects to assume such defense, but the fees and expenses of such counsel shall be at the Indemnified Party's expense.

8.5.4 <u>Settlement of Claims</u>. The Indemnified Party may at any time notify the Indemnifying Party of its intention to settle or compromise any claim, suit or action against the Indemnified Party without the consent of the Indemnifying Party, <u>provided</u> that the Indemnifying Party shall have no further liability in respect thereof.

8.5.5 <u>Subrogation</u>. The Indemnifying Party shall be subrogated to any claims or rights of the Indemnified Party as against any other persons with respect to any amount paid by the Indemnifying Party under this Article 8. The Indemnified Party shall cooperate with the Indemnifying Party, at the Indemnifying Party's expense, in the assertion by the Indemnifying Party of any such claim against such other persons.

8.6 <u>Limitations on Indemnification</u>.

8.6.1 Notwithstanding anything to the contrary set forth in this Article 8, the Buyer Indemnified Parties shall not be entitled to indemnification for any breach of any representation or warranty or covenants made by Seller Parties pursuant to Section 8.1 of this Agreement, unless and until the aggregate amount of all Losses sustained by Buyer Indemnified Parties exceeds an amount equal to \$500,000 (the "Trigger Amount"), it being understood and agreed that once such Losses sustained by the Buyer Indemnified Parties exceed the Trigger Amount, Seller Parties shall be responsible for all Losses incurred by the Buyer Indemnified Parties without regard to the aforementioned Trigger Amount; provided that Seller Parties' maximum liability under this Article 8 shall not exceed \$1,500,000.

8.6.2 Notwithstanding anything to the contrary set forth in this Article 8, Seller Indemnified Parties shall not be entitled to indemnification for any breach of any representation or warranty or covenants made by Buyer Parties pursuant to Section 8.1 of this Agreement, unless and until the aggregate amount of all Losses sustained by Seller Indemnified Parties exceeds the Trigger Amount, it being understood and agreed that once such Losses sustained by the Seller Indemnified Parties exceed the Trigger Amount, Buyer Parties shall be responsible for all Losses incurred by the Seller Indemnified Parties without regard to the aforementioned Trigger Amount; provided that Buyer Parties' maximum liability under this Article 8 shall not exceed \$1,500,000.

8.7 <u>Survival of Indemnification Obligations</u>. The obligations of the Parties set forth in this Article 8 shall survive the Closing until June 30, 2009, except that a claim for indemnification for which notice was given pursuant to Section 8.5 by an Indemnified Party prior to June 30, 2009 shall survive until such claim is fully and finally determined.

8.8 <u>Exclusive Remedy</u>. Absent any fraudulent or intentional misrepresentation, the indemnification provided in this Article 8 shall be the exclusive post-closing remedy available to each Party and its Affiliates and their respective officers, directors and employees in connection with any breach of any representation, warranty, covenant or agreement made by the other party in this Agreement, or for any other matter or claim arising under or in connection with this Agreement, or any of the transactions contemplated hereby.

ARTICLE 9

MISCELLANEOUS PROVISIONS

9.1 <u>Notices</u>. Any notice, request, demand or other communication required or permitted under this Agreement shall be given in writing and shall be delivered or sent by registered or certified mail, return receipt requested in a prepaid envelope, by overnight mail

or courier, or by facsimile transmission, to the addresses set forth below or such other addresses as such party shall hereafter specify in accordance with this Section:

If to Seller:	Elk Associates Funding Corporation 747 Third Avenue New York, NY 10017 Attention: Gary C. Granoff, President Telephone: 212-355-2449 Facsimile: 212-759-3338
If to Ameritrans:	Ameritrans Capital Corporation 747 Third Avenue New York, NY 10017 Attention: Michael Feinsod, President Telephone: 212-355-2449 Facsimile: 212-759-3338
For notices to Seller or Ameritrans, with a copy to (which copy shall not constitute notice):	Paul, Hastings, Janofsky & Walker LLP 75 East 55th Street New York, NY 10022 Attention: Barry A. Brooks Facsimile: 212-230-7777
If to Buyer:	Medallion Bank 1100 East 6600 South, Suite 510 Salt Lake City, Utah 84121 Attention: John Taggart Telephone: 801-284-7060 Facsimile: 801-284-7077 with a copy to MFC at the address below.
If to MFC:	Medallion Financial Corp. 437 Madison Avenue New York, NY 10022 Attention: General Counsel Telephone: 212-328-3615 Facsimile: 212-328-3614

Such notice or other communication shall be deemed to have been given (i) when delivered, if sent by registered or certified mail or delivered personally or by facsimile transmission, or (ii) on the second following Business Day if sent by overnight mail or overnight courier.

9.2 <u>Severability</u>. Each part of this Agreement is intended to be severable. If any term, covenant, condition or provision of this Agreement is held to be unlawful, invalid or

unenforceable by a court of competent jurisdiction, such illegality, invalidity or unenforceability shall not affect the remaining provisions of this Agreement, which shall remain in full force and effect and shall be binding upon the Parties.

9.3 <u>Governing Law</u>. The Parties agree that this Agreement shall be construed, and the rights and obligations of the Parties under the Agreement shall be determined, in accordance with the substantive laws of the State of New York applicable to contracts to be performed entirely within the State of New York as though this Agreement were to be wholly performed within the State of New York and as though all persons who are Parties were domiciliaries of the State of New York.

9.4 <u>Assignments; Binding Effect</u>. Neither Party may assign either this Agreement or any rights or obligations hereunder to any entity other than one of its Affiliates without the prior written approval of the other Party. This Agreement shall inure to the benefit of and be binding upon the respective successors and permitted assigns of the Parties.

9.5 <u>Survival</u>. Seller and Buyer agree that (a) except as expressly limited in this Agreement, the covenants of the Parties herein contained shall survive Closing until 2 years after the Closing Date, shall not merge into the Closing Documents, and shall be independently enforceable, and (b) the representations and warranties of the Parties shall survive Closing until six (6) months after the Closing Date.

9.6 <u>Entire Agreement; Amendments</u>. This Agreement, including any schedules, and exhibits hereto, and the Confidentiality Agreement constitute the entire agreement between the Parties pertaining to the subject matter hereof and supersedes any and all prior agreements, representations and understandings of the Parties, written or oral. The terms of this Agreement shall not be modified or amended except by subsequent written agreement of the Parties.

9.7 <u>Counterparts; Facsimile; PDF</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. Any facsimile or portable document format copies hereof or signature hereon shall, for all purposes, be deemed originals.

9.8 <u>Announcements</u>. Neither Buyer Parties nor Seller Parties shall issue any news release or other public notice or communication without the prior consent of the other Party, except (i) as required or expressly permitted by this Agreement, as may be necessary in order to give the notices to obtain prior regulatory approvals, as necessary to consult with attorneys, accountants, employees, or other advisors retained in connection with the transactions contemplated hereby, (ii) as required by court order or otherwise mandated by law or (iii) in connection with disclosure documents required to be prepared by the Parties or any parent or Affiliate of any of the Parties. Concurrently with the signing of this Agreement, each Party shall be permitted to issue a press release in such form as shall be mutually agreed upon by the Parties.

9.9 <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES KNOWINGLY, VOLUNTARILY AND INTENTIONALLY WAIVES ANY RIGHTS IT MAY HAVE TO A

TRIAL BY JURY IN RESPECT OF ANY LITIGATION BASED HEREON, OR ARISING OUT OF, UNDER, OR IN CONNECTION WITH, THIS AGREEMENT OR ANY SCHEDULE HERETO, OR ANY COURSE OF CONDUCT, COURSE OF DEALING OR STATEMENTS (WHETHER VERBAL OR WRITTEN) RELATING TO THE FOREGOING. THIS PROVISION IS A MATERIAL INDUCEMENT FOR THE PARTIES TO ENTER INTO THIS AGREEMENT.

9.10 <u>Consent to Jurisdiction</u>. The Parties hereby submit to the nonexclusive jurisdiction of the federal and state courts of New York County, New York, in connection with any action arising out of or relating to this Agreement and any of the agreements, instruments and other documents entered into in connection herewith and the Parties hereby irrevocably agree that all claims in respect of any such suit, action or proceeding shall be heard and determined in such courts. The Parties hereby irrevocably waive any objection which they may now or hereafter have to the laying of venue of any such suit, action or proceeding brought in such court or any defense of inconvenient forum for the maintenance of any such suit, action or proceeding. Each of the Parties agrees that a judgment in any such suit, action or proceeding may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law.

9.11 <u>No Third Party Beneficiary</u>. This Agreement is not intended and shall not be construed to confer upon any Person other than the Parties any rights or remedies hereunder.

9.12 <u>Headings; References; Interpretation</u>. The headings of the Articles, Sections and subsections of this Agreement are solely for convenience and reference and shall not limit or otherwise affect the meaning or interpretation of any of the terms or provisions of this Agreement. The references herein to Articles, Sections and subsections, and Schedules, unless otherwise indicated, are references to sections and subsections of and schedules to this Agreement. Whenever the words "*include*", "*includes*", or "*including*" are used in this Agreement, they shall be deemed to be followed by the words "*without limitation*". Unless the context requires otherwise, all words used in this Agreement in the singular number shall extend to and include the plural number, all words in any gender shall extend to and include all genders. This Agreement shall be construed without regard to any presumption or rule requiring construction against the party drafting or causing this instrument to be drafted.

9.13 <u>Waiver</u>. No waiver by a Party of another Party's breach of any term, covenant or condition contained in this Agreement shall be deemed to be a waiver of any subsequent breach of the same or any other term, covenant or condition of this Agreement. No failure or delay by any Party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. Except as otherwise set forth herein, the rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by law.

9.14 <u>Currency Designation</u>. Except as expressly provided herein, all amounts specified in this Agreement shall be in United States dollars.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed and sealed as of the day and year first written above.

AMERITRANS:

AMERITRANS CAPITAL CORPORATION

By:_____ Name: Michael Feinsod Title: President

BUYER:

MEDALLION BANK

By:	
Name:	
Title:	

MFC:

MEDALLION FINANCIAL CORP.

By:	
Name:	
Title:	

SELLER:

ELK ASSOCIATES FUNDING CORPORATION

By:_____

Name: Gary C. Granoff Title: President

Loan Number	Paid Thru Date	Loan Balance	Interest Due	Funding Date	No. Pmts	Rate	Original Balance	Loan Type	Monthly Payment	
Boston		~								
Loans		n, Cambridge and S								
EAF043780	06/01/2008	100,221.80	0.00	02/06/2003	144	7.25	124,500.00	Balloon	1,138.55	
EAF043779	06/01/2008	129,488.44	0.00	02/06/2003	176	7.25	162,500.00	Balloon	1,486.06	
EAF043841	06/01/2008	127,807.67	0.00	05/07/2003	177	7.00	165,000.00	Balloon	1,488.25	
EAF043842	06/01/2008	45,691.43	0.00	05/07/2003	179	8.00	51,000.00	Balloon	453.98	
EAF044295	06/01/2008	215,030.62	0.00	03/16/2006	180	7.50	235,000.00	Balloon	2,178.48	
EAF044325	06/01/2008	166,598.33	0.00	06/07/2006	180	7.50	180,000.00	Balloon	1,668.62	
EAF044347	06/01/2008	249,474.80	0.00	09/21/2006	228	7.90	260,000.00	Balloon	2,205.77	
EAF044348	06/01/2008	249,474.80	0.00	09/21/2006	228	7.90	260,000.00	Balloon	2,205.77	
EAF044356	06/01/2008	228,855.69	0.00	11/01/2006	180	7.90	243,500.00	Balloon	2,312.98	
EAF044382	06/01/2008	209,524.35	0.00	01/31/2007	252	8.13	215,000.00	Balloon	1,780.91	
EAF044403	05/01/2008	293,604.06	0.00	03/01/2007	180	8.13	305,000.00	Balloon	2,936.79	
EAF044415	04/01/2008	248,298.22	1,255.01	04/09/2007	264	8.25	250,000.00	Balloon	2,055.56	
EAF044430	06/01/2008	68,588.62	0.00	05/02/2007	144	8.25	73,500.00	Balloon	805.71	
EAF044457	06/01/2008	43,984.75	0.00	06/07/2007	144	8.25	46,300.00	Balloon	507.54	
EAF044458	06/01/2008	45,412.22	0.00	06/07/2007	144	8.25	47,700.00	Balloon	522.89	
EAF044456	06/01/2008	240,409.32	0.00	06/07/2007	240	8.25	245,000.00	Balloon	2,087.56	
EAF044465	06/01/2008	96,849.52	0.00	06/26/2007	180	8.50	100,000.00	Balloon	984.74	
EAF044467	06/01/2008	43,646.17	0.00	06/27/2007	120	8.25	46,500.00	Balloon	570.33	
EAF044471	07/01/2008	36,805.53	0.00	06/28/2007	99	8.25	40,000.00	Balloon	555.56	
EAF044516	06/01/2008	320,248.55	0.00	10/16/2007	300	8.75	322,500.00	Balloon	2,651.41	
EAF044524	06/01/2008	312,347.73	0.00	10/26/2007	276	8.50	315,000.00	Balloon	2,602.17	
EAF044525	06/01/2008	66,194.64	0.00	11/01/2007	180	8.75	67,500.00	Balloon	674.63	
EAF044533	06/01/2008	295,019.37	0.00	11/15/2007	216	8.75	298,500.00	Balloon	2,748.86	
EAF044547	06/01/2008	204,840.93	0.00	12/20/2007	288	8.25	206,000.00	Balloon	1,644.90	
EAF044555	06/01/2008	179,881.51	0.00	12/28/2007	216	9.00	181,600.00	Balloon	1,700.58	
EAF044562	06/01/2008	338,709.96	0.00	02/06/2008	264	8.75	340,000.00	Balloon	2,906.06	
Total: Boston		4,557,009.03							_,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
	_									
EAF053047	06/01/2008	29,541.63	0.00	12/03/1999	60	10.90	60,000.00	Regular	989.10	

Schedule A SCHEDULE OF ACQUIRED LOANS

Loan	Paid Thru	T D I	Interest	Funding	No.		Original	.	Monthly
Number EAF053062	Date 05/01/2008	Loan Balance 29,561.60	Due 0.00	Date 12/15/1999	Pmts 72	Rate 11.90	Balance 60,000.00	Loan Type	Payment 937.06
		,						Regular	
EAF053140	05/01/2008	19,548.91	170.70	03/17/2000	72	10.75	60,000.00	Regular	1,073.27
EAF053227	05/01/2008	39,526.23	0.00	07/28/2000	138	12.00	52,000.00	Balloon	591.75
EAF053320	05/01/2008	33,016.21	0.00	12/11/2000	120	11.00	60,000.00	Balloon	669.98
EAF053349	05/01/2008	38,088.19	333.27	02/07/2001	120	10.50	62,500.00	Balloon	828.64
EAF053597	08/01/2008	34,047.68	0.00	05/08/2002	130	11.50	58,500.00	Regular	757.72
EAF053655	06/01/2008	23,305.54	0.00	08/21/2002	60	11.00	57,000.00	Regular	763.00
EAF053689	07/01/2008	24,800.95	0.00	09/26/2002	104	11.50	58,500.00	Regular	860.96
EAF053707	05/01/2008	40,078.92	56.95	10/25/2002	120	11.00	44,000.00	Balloon	675.65
EAF053713	06/01/2008	33,679.44	0.00	11/19/2002	48	8.75	60,000.00	Regular	988.92
EAF053760	06/01/2008	19,259.12	0.00	01/15/2003	84	10.50	60,000.00	Regular	1,011.64
EAF053794	06/01/2008	1,540.09	0.00	02/19/2003	31	11.00	20,000.00	Regular	434.85
EAF053871	07/01/2008	17,008.01	0.00	09/11/2003	84	11.50	40,000.00	Regular	695.46
EAF053887	06/01/2008	143,421.44	0.00	10/31/2003	144	6.00	205,000.00	Balloon	2,000.49
EAF053893	06/01/2008	137,462.37	0.00	10/31/2003	144	6.00	205,000.00	Balloon	2,000.49
EAF053899	05/01/2008	145,969.08	729.85	11/30/2003	144	6.00	205,000.00	Balloon	2,000.49
EAF053929	05/01/2008	42,453.91	406.85	01/16/2004	120	11.50	59,000.00	Balloon	687.22
EAF053948	06/01/2008	46,207.44	0.00	01/20/2004	180	5.25	58,500.00	Balloon	470.27
EAF053936	06/01/2008	27,471.23	0.00	01/26/2004	84	10.75	59,000.00	Regular	1,007.23
EAF053996	06/01/2008	46,718.65	0.00	04/12/2004	180	5.25	58,500.00	Balloon	470.27
EAF053997	06/01/2008	46,658.46	0.00	04/15/2004	180	5.25	58,500.00	Balloon	470.27
EAF054027	06/01/2008	47,557.54	0.00	06/25/2004	180	5.25	58,500.00	Balloon	470.27
EAF054048	06/01/2008	583,312.95	0.00	08/31/2004	180	5.00	711,600.00	Regular	5,627.29
EAF054065	05/13/2008	9,703.93	42.34	10/13/2004	180	5.25	11,700.00	Balloon	94.05
EAF054074	06/01/2008	29,569.54	0.00	10/29/2004	96	11.00	45,000.00	Regular	706.88
EAF054084	06/01/2008	24,447.01	0.00	12/03/2004	72	9.95	48,500.00	Regular	897.28
EAF054086	05/13/2008	9,810.17	42.79	12/13/2004	180	5.25	11,700.00	Balloon	94.05
EAF054103	10/01/2008	11,818.40	0.00	12/20/2004	54	11.50	36,154.03	Regular	833.84
EAF054102	10/01/2008	6,121.12	0.00	12/20/2004	46	12.50	36,683.52	Regular	983.48
EAF054102	11/01/2008	27,553.56	0.00	12/20/2004	84	10.00	44,298.37	Regular	713.44
EAF054110	10/01/2008	13,321.91	0.00	12/20/2004	56	11.50	37,169.59	Regular	833.80
EAF054115	10/01/2008	28,697.57	0.00	12/29/2004	30 84	9.00	46,852.34	Regular	731.24
EAF054115 EAF054116	10/01/2008	37,971.70	0.00	12/29/2004	84 124	9.00 9.00	40,852.54 48,306.10	Regular	578.52
EAF054110 EAF054117	10/01/2008	37,885.69	0.00	12/29/2004	124	9.00 9.00	48,306.10	-	576.00
ЕАГОЈ411/	10/01/2008	57,005.09	0.00	12/29/2004	123	9.00	40,134.11	Regular	370.00

Loan	Paid Thru		Interest	Funding	No.	_	Original		Monthly
Number	Date	Loan Balance	Due	Date	Pmts	Rate	Balance	Loan Type	Payment
EAF054118	10/01/2008	37,775.02	0.00	12/29/2004	124	9.00	48,057.17	Regular	575.56
EAF054119	10/01/2008	28,917.74	0.00	12/29/2004	84	9.00	47,212.82	Regular	736.88
EAF054120	10/01/2008	28,763.87	0.00	12/29/2004	84	9.00	46,961.62	Regular	732.96
EAF054121	10/01/2008	38,024.55	0.00	12/29/2004	124	9.00	48,418.17	Regular	580.60
EAF054122	10/01/2008	37,947.44	0.00	12/29/2004	124	9.00	48,320.61	Regular	579.44
EAF054123	10/01/2008	38,127.41	0.00	12/29/2004	124	9.00	48,550.19	Regular	582.20
EAF054124	10/01/2008	37,538.84	0.00	12/29/2004	124	9.00	47,755.27	Regular	571.92
EAF054125	10/01/2008	28,853.07	0.00	12/29/2004	85	9.00	46,982.27	Regular	731.68
EAF054126	10/01/2008	37,806.31	0.00	12/29/2004	124	9.00	48,097.09	Regular	576.04
EAF054127	10/01/2008	28,617.61	0.00	12/29/2004	85	9.00	46,600.68	Regular	725.76
EAF054128	10/01/2008	28,713.27	0.00	12/29/2004	84	9.00	46,879.36	Regular	731.68
EAF054129	10/01/2008	37,606.68	0.00	12/29/2004	124	9.00	47,843.17	Regular	573.00
EAF054130	10/01/2008	37,279.43	0.00	12/29/2004	124	9.00	47,470.00	Regular	569.24
EAF054132	10/01/2008	29,123.56	0.00	12/29/2004	84	9.00	47,548.68	Regular	742.12
EAF054133	10/01/2008	37,700.98	0.00	12/29/2004	124	9.00	48,005.51	Regular	575.64
EAF054135	10/01/2008	37,636.27	0.00	12/29/2004	124	9.00	47,880.46	Regular	573.44
EAF054136	10/01/2008	37,317.95	0.00	12/29/2004	124	9.00	47,474.45	Regular	568.56
EAF054157	11/01/2008	10,106.38	0.00	01/24/2005	48	13.00	41,523.39	Regular	1,081.52
EAF054167	10/01/2008	5,780.63	0.00	01/27/2005	45	10.00	36,949.56	Regular	967.68
EAF054171	09/01/2008	23,267.06	0.00	01/28/2005	68	11.75	44,299.26	Regular	865.84
EAF054180	08/01/2008	27,587.58	0.00	02/03/2005	78	10.50	45,000.00	Regular	781.52
EAF054183	08/01/2008	6,036.87	0.00	02/07/2005	44	12.25	37,047.24	Regular	1,029.52
EAF054192	06/01/2008	119.78	0.00	02/18/2005	36	9.80	20,000.00	Regular	643.47
EAF054206	10/01/2008	22,301.53	0.00	03/17/2005	62	11.75	46,163.41	Regular	973.40
EAF054208	09/01/2008	20,196.56	0.00	03/23/2005	64	11.75	39,524.27	Regular	809.32
EAF054217	06/01/2008	13,021.93	0.00	04/22/2005	60	11.00	29,000.00	Regular	630.53
EAF054219	06/01/2008	12,432.64	0.00	04/29/2005	60	11.00	28,000.00	Regular	608.79
EAF054241	06/01/2008	272,479.71	0.00	08/09/2005	219	6.50	304,500.00	Balloon	2,500.00
EAF054242	06/01/2008	272,479.71	0.00	08/09/2005	219	6.50	304,500.00	Balloon	2,500.00
EAF054242	06/01/2008	272,479.71	0.00	08/09/2005	219	6.50	304,500.00	Balloon	2,500.00
EAF054243	06/01/2008	272,479.71	0.00	08/09/2005	219	6.50	304,500.00	Balloon	2,500.00
EAF054257	07/01/2008	30,748.18	0.00	10/24/2005	84	12.00	42,500.00	Regular	750.24
EAF034237 EAF054269	06/01/2008	43,720.05	0.00	01/09/2006	84 84	12.00	42,300.00 58,500.00	-	1,001.66
	06/01/2008		0.00					Regular	812.97
EAF054278	00/01/2008	29,999.38	0.00	02/09/2006	72	10.75	43,000.00	Balloon	812.97

Loan	Paid Thru		Interest	Funding	No.		Original		Monthly
Number	Date	Loan Balance	Due	Date	Pmts	Rate	Balance	Loan Type	Payment
EAF054279	06/01/2008	268,627.22	0.00	02/14/2006	273	6.50	304,500.00	Balloon	2,500.00
EAF054291	06/01/2008	0.00	0.00	03/09/2006	20	13.00	5,000.00	Regular	276.16
EAF054309	06/01/2008	31,107.65	0.00	04/25/2006	72	10.75	43,000.00	Balloon	812.97
EAF054321	08/01/2008	9,091.67	0.00	05/25/2006	39	10.50	20,000.00	Regular	598.32
EAF054324	06/01/2008	278,925.20	0.00	05/31/2006	121	6.50	304,500.00	Balloon	2,500.00
EAF054328	06/01/2008	452,063.93	0.00	06/12/2006	144	5.00	519,750.00	Balloon	4,764.29
EAF054345	05/01/2008	45,442.66	0.00	09/12/2006	120	10.75	50,000.00	Balloon	681.69
EAF054346	06/01/2008	41,111.26	0.00	09/14/2006	84	10.25	50,000.00	Balloon	836.53
EAF054350	06/01/2008	44,770.15	0.00	10/16/2006	72	10.25	56,500.00	Regular	1,053.85
EAF054355	06/01/2008	41,868.01	0.00	10/30/2006	72	11.25	52,500.00	Regular	1,006.02
EAF054359	06/01/2008	12,013.53	0.00	11/06/2006	48	9.25	18,000.00	Regular	450.07
EAF054383	06/01/2008	45,463.70	0.00	02/01/2007	84	10.25	52,850.00	Balloon	884.21
EAF054385	06/01/2008	235,778.72	0.00	02/05/2007	180	8.00	247,000.00	Balloon	2,360.46
EAF054386	06/01/2008	235,778.72	0.00	02/05/2007	180	8.00	247,000.00	Balloon	2,360.46
EAF054387	06/01/2008	176,833.97	0.00	02/05/2007	180	8.00	185,250.00	Balloon	1,770.35
EAF054388	06/01/2008	235,778.72	0.00	02/05/2007	180	8.00	247,000.00	Balloon	2,360.46
EAF054389	06/01/2008	235,778.72	0.00	02/05/2007	180	8.00	247,000.00	Balloon	2,360.46
EAF054390	06/01/2008	235,778.72	0.00	02/05/2007	180	8.00	247,000.00	Balloon	2,360.46
EAF054391	06/01/2008	235,778.72	0.00	02/05/2007	180	8.00	247,000.00	Balloon	2,360.46
EAF054397	06/01/2008	43,817.33	0.00	02/14/2007	60	10.50	55,000.00	Regular	1,182.16
EAF054398	06/01/2008	96,672.46	0.00	02/16/2007	84	12.50	110,000.00	Regular	1,971.34
EAF054406	06/01/2008	56,680.40	0.00	03/20/2007	84	10.25	64,500.00	Balloon	1,079.13
EAF054420	06/01/2008	34,317.85	0.00	04/18/2007	48	8.75	45,000.00	Regular	1,114.49
EAF054423	08/01/2008	638,767.72	0.00	04/19/2007	180	10.00	652,800.00	Balloon	6,466.48
EAF054424	08/01/2008	718,613.56	0.00	04/19/2007	180	10.00	734,400.00	Balloon	7,274.80
EAF054428	05/01/2008	18,764.86	0.00	05/01/2007	36	8.75	26,000.00	Regular	823.77
EAF054431	06/01/2008	30,392.33	0.00	05/03/2007	72	9.00	35,000.00	Balloon	630.89
EAF054438	06/01/2008	64,416.12	0.00	05/16/2007	240	10.00	65,500.00	Balloon	632.09
EAF054439	06/01/2008	64,416.12	0.00	05/16/2007	240	10.00	65,500.00	Balloon	632.09
EAF054440	06/01/2008	64,416.12	0.00	05/16/2007	240	10.00	65,500.00	Balloon	632.09
EAF054441	06/01/2008	64,416.12	0.00	05/16/2007	240	10.00	65,500.00	Balloon	632.09
EAF054442	06/01/2008	64,416.12	0.00	05/16/2007	240	10.00	65,500.00	Balloon	632.09
EAF054443	06/01/2008	64,416.12	0.00	05/16/2007	240	10.00	65,500.00	Balloon	632.09
EAF054444	06/01/2008	64,416.12	0.00	05/16/2007	240	10.00	65,500.00	Balloon	632.09
							,		

Loan	Paid Thru		Interest	Funding	No.		Original		Monthly
Number	Date 06/01/2008	Loan Balance	Due 0.00	Date 05/16/2007	Pmts	Rate 10.00	Balance	Loan Type	Payment 632.09
EAF054445 EAF054446	06/01/2008	64,416.12 64,417.01	0.00	05/16/2007	240 240	10.00	65,500.00 65,500.00	Balloon Balloon	632.09
		0.00					,		
EAF054461	07/01/2008		0.00	06/11/2007	120	10.95	63,750.00	Balloon	876.35
EAF054464	06/01/2008	61,013.53	0.00	06/15/2007	110	8.00	64,500.00	Balloon	789.34
EAF054469	06/01/2008	61,772.48	0.00	06/27/2007	120	9.00	63,000.00	Balloon	798.06
EAF054481	06/01/2008	73,096.93	0.00	07/12/2007	120	10.00	77,000.00	Balloon	1,017.56
EAF054483	06/01/2008	55,056.40	0.00	07/13/2007	36	9.00	75,000.00	Regular	2,384.98
EAF054494	06/01/2008	46,588.46	0.00	08/15/2007	120	8.50	49,000.00	Balloon	607.53
EAF054499	06/01/2008	75,649.80	0.00	08/21/2007	120	10.00	79,250.00	Balloon	1,047.29
EAF054504	06/01/2008	69,142.13	0.00	08/29/2007	240	10.00	70,000.00	Balloon	675.52
EAF054510	06/01/2008	198,005.86	0.00	09/20/2007	240	8.75	200,000.00	Balloon	1,767.42
EAF054515	06/01/2008	76,959.43	0.00	10/05/2007	120	8.50	80,000.00	Balloon	991.89
EAF054514	06/01/2008	71,354.12	0.00	10/05/2007	120	10.00	75,000.00	Balloon	991.13
EAF054521	06/01/2008	54,792.24	0.00	10/18/2007	76	9.00	58,500.00	Regular	990.56
EAF054526	06/01/2008	83,320.01	0.00	11/01/2007	180	8.50	85,000.00	Balloon	837.03
EAF054527	06/01/2008	62,529.54	0.00	11/01/2007	120	8.50	65,000.00	Balloon	805.91
EAF054528	06/01/2008	73,549.59	0.00	11/01/2007	180	8.75	75,000.00	Balloon	749.59
EAF054535	06/01/2008	83,626.70	0.00	11/27/2007	180	9.00	85,000.00	Balloon	862.13
EAF054537	05/01/2008	83,808.51	591.58	11/30/2007	180	8.50	85,000.00	Balloon	837.03
EAF054538	05/01/2008	75,692.79	0.00	11/30/2007	120	8.50	77,000.00	Balloon	954.69
EAF054540	06/01/2008	16,688.77	0.00	12/06/2007	60	8.25	40,000.00	Regular	815.85
EAF054541	06/01/2008	72,978.36	0.00	12/10/2007	120	8.50	75,000.00	Balloon	929.89
EAF054545	05/01/2008	76,384.94	363.23	12/14/2007	180	9.00	77,000.00	Balloon	780.99
EAF054544	06/01/2008	285,000.00	0.00	12/14/2007	180	9.00	285,000.00	Interest Only	2,137.50
EAF054550	05/01/2008	49,071.96	377.47	12/24/2007	120	9.25	50,000.00	Regular	640.16
EAF054553	06/01/2008	59,272.57	0.00	12/27/2007	84	8.75	62,000.00	Regular	989.67
EAF054554	06/01/2008	85,000.00	0.00	12/28/2007	239	9.00	85,000.00	Interest Only	637.50
EAF054557	06/01/2008	59,358.59	0.00	01/14/2008	180	9.00	60,000.00	Balloon	608.56
EAF054559	05/01/2008	77,923.11	584.42	01/17/2008	84	9.00	80,000.00	Balloon	1,287.13
EAF054560	06/01/2008	94,152.69	0.00	01/25/2008	180	11.00	95,000.00	Balloon	1,079.77
EAF054561	06/01/2008	84,290.15	0.00	02/04/2008	180	8.50	85,000.00	Balloon	837.03
EAF054563	06/01/2008	418,080.06	0.00	02/06/2008	120	8.25	425,000.00	Balloon	5,212.74
EAF054565	06/01/2008	39,232.66	0.00	02/08/2008	108	8.00	40,000.00	Balloon	520.75
EAF054569	06/01/2008	68,906.65	0.00	02/28/2008	120	9.00	70,000.00	Balloon	886.73
L/H 05-507	00/01/2000	00,700.05	0.00	02/20/2000	120	2.00	70,000.00	Duntoon	000.75

Loan	Paid Thru		Interest	Funding	No.	_	Original		Monthly			
Number	Date	Loan Balance	Due	Date	Pmts	Rate	Balance	Loan Type	Payment	r	1	
EAF054571	06/01/2008	84,118.23	0.00	03/10/2008	120	9.00	85,000.00	Regular	1,076.74	1		4/1/2018
EAF054572	06/01/2008	56,479.49	0.00	03/13/2008	41	8.00	59,000.00	Regular	1,649.40	1		9/1/2011
EAF054574	06/01/2008	72,198.88	0.00	03/18/2008	120	8.00	74,000.00	Balloon	897.82	1		4/1/2013
EAF054576	06/01/2008	78,393.31	0.00	03/18/2008	120	8.00	80,000.00	Balloon	970.62	1		4/1/2013
EAF054575	06/01/2008	83,492.71	0.00	03/18/2008	84	8.25	85,000.00	Regular	1,335.44	1		4/1/2015
EAF054577	06/01/2008	91,245.39	0.00	03/19/2008	144	8.25	92,000.00	Balloon	1,008.51	1		4/1/2013
EAF054580	06/01/2008	85,004.19	0.00	03/24/2008	180	8.00	85,500.00	Balloon	817.08	1		4/1/2013
EAF054579	06/01/2008	154,120.80	0.00	03/24/2008	180	8.25	155,000.00	Balloon	1,503.72	2		4/1/2013
EAF054582	06/01/2008	74,565.08	0.00	03/26/2008	180	8.00	75,000.00	Balloon	716.74	1		4/1/2013
EAF054583	06/01/2008	89,745.62	0.00	04/08/2008	180	8.25	90,000.00	Balloon	873.13	1		5/1/2013
EAF054585	06/01/2008	84,754.37	0.00	04/17/2008	180	8.00	85,000.00	Balloon	812.30	1		5/1/2013
EAF054588	06/01/2008	199,422.03	0.00	04/25/2008	180	8.00	200,000.00	Balloon	1,911.30	2		5/1/2013
EAF054587	06/01/2008	49,730.49	0.00	04/25/2008	120	8.25	50,000.00	Balloon	613.26	1		5/1/2013
EAF054589	06/01/2008	80,000.00	0.00	05/02/2008	180	8.00	80,000.00	Balloon	764.52	1		6/1/2013
EAF054592	06/01/2008	82,000.00	0.00	05/09/2008	180	8.00	82,000.00	Balloon	783.63	1		6/1/2013
EAF054590	06/01/2008	91,000.00	0.00	05/09/2008	24	9.00	91,000.00	Regular	4,157.31		Car Loan - Secured by liens on 11	
										*	medallions	6/1/2010
EAF054596	06/01/2008	71,000.00	0.00	05/15/2008	120	8.00	71,000.00	Balloon	861.43	1		6/1/2013
EAF054597	06/01/2008	150,000.00	0.00	05/19/2008	180	8.00	180,000.00	Balloon	1,720.17	2		6/1/2013
EAF054598	06/01/2008	68,000.00	0.00	05/19/2008	120	8.00	68,000.00	Balloon	825.03	1		6/1/2013
EAF054599	06/01/2008	100,000.00	266.67	05/20/2008	180	8.00	100,000.00	Balloon	955.65	1		6/1/2013
EAF054603	06/01/2008	50,620.00	0.00	06/27/2008	84	8.00	50,620.00	Balloon	788.97	19		7/1/2013
EAF054604	06/01/2008	45,000.00	0.00	06/27/2008	144	8.25	45,000.00	Balloon	493.29	4		7/1/2013
EAF054605	06/01/2008	348,000.00	0.00	06/27/2008	144	8.25	348,000.00	Balloon	3,814.80	3		7/1/2013
EAF054606	06/01/2008	232,000.00	0.00	06/27/2008	144	8.25	232,000.00	Balloon	2,543.20	2		7/1/2013
Total: Chicago	_	14,203,076.97										

Florida Loans

EAF063697 06/01/2008 28,459.32 0.00 10/11/2002 92 14.50 66,500.00 Regular 1,300.00 EAF063846 06/01/2008 31,813.95 0.00 05/20/2003 131 10.00 59,328.00 Balloon 745.99 EAF063860 06/01/2008 60,194.73 0.00 07/23/2003 203 13.00 67,450.00 Balloon 904.00 EAF063908 06/01/2008 52,850.67 0.00 10/10/2003 87 13.00 17,575.00 Balloon 1,164.50 EAF063909 06/01/2008 9,42.96 0.00 12/11/2003 98 13.00 80,100.00 Balloon 1,516.67 EAF064090 06/01/2008 61,460.94 0.00 12/11/2003 111 14.00 87,400.00 Balloon 1,500.00 EAF0640404 06/01/2008 64,022.04 0.00 08/05/2004 141 14.00 87,400.00 Balloon 1,25.00 EAF064053 06/01/2008 78,755.22 0.00 11/2/2004
EAF06384606/01/200831,813.950.0005/20/200313110.0059,328.00Balloon745.90EAF06386006/01/200860,194.730.0007/23/200320313.0067,450.00Balloon904.00EAF06389206/01/200852,850.670.0010/10/200312014.0071,250.00Balloon1,164.50EAF06390806/01/20089,642.960.0012/11/20038713.0017,575.00Balloon325.00EAF06390606/01/200861,460.940.0012/11/20039813.0080,100.00Balloon1,516.67EAF06400406/01/200864,009.520.0004/27/200311114.0087,400.00Balloon1,500.00EAF06405306/01/200864,022.040.0008/05/200414114.0073,340.00Balloon1,130.00EAF06405306/01/200852,570.950.0009/15/200411415.0065,075.00Balloon1,230.00EAF06405206/01/200878,795.220.0001/21/200521512.5085,500.00Balloon1,300.00EAF06422606/01/200878,539.520.0005/02/200517412.00101,650.00Balloon1,300.00EAF06422606/01/200878,539.520.0005/02/200517412.00101,650.00Balloon1,300.00EAF06422606/01/200813,170.980.0005/02/20058411.5024,510.00Regu
EAF06386006/01/200860,194.730.0007/23/200320313.0067,450.00Balloon904.00EAF06389206/01/200852,850.670.0010/10/200312014.0071,250.00Balloon1,164.50EAF06390806/01/20089,642.960.0012/11/20038713.0017,575.00Balloon325.00EAF06390906/01/200849,720.770.0012/11/20039813.0080,100.00Balloon1,516.6'EAF06390606/01/200861,460.940.0012/11/200311114.0080,820.00Balloon1,437.00EAF06400406/01/200864,009.520.0004/27/200410814.0087,400.00Balloon1,516.6'EAF06405306/01/200864,022.040.0008/05/200414114.0073,340.00Balloon1,130.00EAF06405306/01/200878,795.220.0001/12/200412013.5095,000.00Balloon1,52.7'EAF06415606/01/200878,595.220.0001/21/200521512.5085,500.00Balloon1,300.00EAF06422306/01/200878,539.520.0005/02/200517412.00101,650.00Balloon1,300.00EAF06425406/01/200813,170.980.0005/02/200517412.00101,650.00Balloon2,95.22EAF06425406/01/200819,482.590.0011/05/20052849.00257,642.00Bal
EAF06389206/01/200852,850.670.0010/10/200312014.0071,250.00Balloon1,164.50EAF06390806/01/20089,642.960.0012/11/20038713.0017,575.00Balloon325.00EAF06390906/01/200849,720.770.0012/11/20039813.0080,100.00Balloon1,516.67EAF06390606/01/200861,460.940.0012/11/200311114.0080,820.00Balloon1,437.00EAF06400406/01/200864,009.520.0004/27/200410814.0087,400.00Balloon1,500.00EAF06405306/01/200864,022.040.0008/05/200414114.0073,340.00Balloon1,130.00EAF06405206/01/200878,795.220.0001/21/200412013.5095,000.00Balloon1,522.77EAF06415606/01/200878,539.520.0001/21/200521512.5085,500.00Balloon1,300.00EAF06422306/01/200878,539.520.0005/02/200517412.00101,650.00Balloon1,300.00EAF06422606/01/200873,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/200819,482.590.0010/05/20052849.00257,642.00Balloon2,195.24EAF06426406/01/200810,5195.790.0012/01/200512213.0077,235.00Ba
EAF06390806/01/20089,642.960.0012/11/20038713.0017,575.00Balloon325.00EAF06390906/01/200849,720.770.0012/11/20039813.0080,100.00Balloon1,516.67EAF06390606/01/200861,460.940.0012/11/200311114.0080,820.00Balloon1,437.00EAF06400406/01/200864,009.520.0004/27/200410814.0087,400.00Balloon1,500.00EAF06405306/01/200864,022.040.0008/05/200414114.0073,340.00Balloon1,125.00EAF06405306/01/200852,570.950.0009/15/200411415.0065,075.00Balloon1,300.00EAF06408206/01/200878,795.220.0011/2/2/200412013.5095,000.00Balloon1,520.00EAF06415606/01/200878,539.520.0001/21/200521512.5085,500.00Balloon1,300.00EAF06422306/01/200897,092.320.0005/02/200517412.00101,650.00Balloon1,300.00EAF06425406/01/200813,170.980.0001/05/20058413.0024,985.00Balloon2,195.22EAF06425406/01/200819,482.590.0010/05/20058413.0024,985.00Balloon2,195.22EAF06425406/01/200810,948.590.0012/01/200512213.0077,235.00B
EAF06390906/01/200849,720.770.0012/11/20039813.0080,100.00Balloon1,516.67EAF06390606/01/200861,460.940.0012/11/200311114.0080,820.00Balloon1,437.00EAF06400406/01/200864,009.520.0004/27/200410814.0087,400.00Balloon1,500.00EAF06404106/01/200864,022.040.0008/05/200414114.0073,340.00Balloon1,125.00EAF06405306/01/200852,570.950.0009/15/200411415.0065,075.00Balloon1,300.00EAF06415606/01/200878,795.220.0011/22/200412013.5095,000.00Balloon1,522.74EAF06415606/01/200878,539.520.0001/21/200521512.5085,500.00Balloon1,300.00EAF06422306/01/200879,092.320.0005/02/200517412.00101,650.00Balloon1,300.00EAF06425406/01/200813,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/200819,482.590.0011/03/20058413.0024,985.00Balloon1,200.00EAF06426206/01/2008105,195.790.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00<
EAF06390606/01/200861,460.940.0012/11/200311114.0080,820.00Balloon1,437.00EAF06400406/01/200864,009.520.0004/27/200410814.0087,400.00Balloon1,500.00EAF06404106/01/200864,022.040.0008/05/200414114.0073,340.00Balloon1,125.00EAF06405306/01/200852,570.950.0009/15/200411415.0065,075.00Balloon1,130.00EAF06408206/01/200878,795.220.0011/22/200412013.5095,000.00Balloon1,522.74EAF06415606/01/200878,539.520.0001/21/200521512.5085,500.00Balloon1,300.00EAF06422306/01/200897,092.320.0005/02/200517412.00101,650.00Balloon1,300.00EAF06425406/01/200813,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/200819,482.590.0011/03/20052849.00257,642.00Balloon1,200.00EAF06426206/01/200819,482.590.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,200.00EAF06426606/01/2008105,195.790.0012/08/200534212.00101,080.00
EAF06400406/01/200864,009.520.0004/27/200410814.0087,400.00Balloon1,500.00EAF06404106/01/200864,022.040.0008/05/200414114.0073,340.00Balloon1,125.00EAF06405306/01/200852,570.950.0009/15/200411415.0065,075.00Balloon1,130.00EAF06408206/01/200878,795.220.0011/22/200412013.5095,000.00Balloon1,522.74EAF06415606/01/200878,539.520.0001/21/200521512.5085,500.00Balloon936.00EAF06422306/01/200897,092.320.0005/02/200517412.00101,650.00Balloon1,300.00EAF06425406/01/200813,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/200819,482.590.0011/03/20052849.00257,642.00Balloon1,200.00EAF06426206/01/200819,482.590.0011/03/20058413.0024,985.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426606/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,00.00EAF06426606/01/200887,729.260.0002/08/200611412.0097,280.00 <td< td=""></td<>
EAF06404106/01/200864,022.040.0008/05/200414114.0073,340.00Balloon1,125.00EAF06405306/01/200852,570.950.0009/15/200411415.0065,075.00Balloon1,130.00EAF06408206/01/200878,795.220.0011/22/200412013.5095,000.00Balloon1,522.77EAF06415606/01/200878,539.520.0001/21/200521512.5085,500.00Balloon1,300.00EAF06422306/01/200897,092.320.0005/02/200517412.00101,650.00Balloon1,300.00EAF06425406/01/200813,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/2008248,502.890.0011/03/20052849.00257,642.00Balloon2,195.23EAF06426206/01/200819,482.590.0011/03/20058413.0024,985.00Balloon478.43EAF06426406/01/200810,5195.790.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426606/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,00.00EAF06426606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00 <t< td=""></t<>
EAF06405306/01/200852,570.950.0009/15/200411415.0065,075.00Balloon1,130.00EAF06408206/01/200878,795.220.0011/22/200412013.5095,000.00Balloon1,522.74EAF06415606/01/200878,539.520.0001/21/200521512.5085,500.00Balloon936.00EAF06422306/01/200897,092.320.0005/02/200517412.00101,650.00Balloon1,300.00EAF06422606/01/200813,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/2008248,502.890.0010/05/20052849.00257,642.00Balloon2,195.22EAF06426206/01/200819,482.590.0011/03/20058413.0024,985.00Balloon478.42EAF06426406/01/2008105,195.790.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426606/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,100.00EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,600.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50
EAF06408206/01/200878,795.220.0011/22/200412013.5095,000.00Balloon1,522.74EAF06415606/01/200878,539.520.0001/21/200521512.5085,500.00Balloon936.00EAF06422306/01/200897,092.320.0005/02/200517412.00101,650.00Balloon1,300.00EAF06422606/01/200813,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/2008248,502.890.0010/05/20052849.00257,642.00Balloon2,195.23EAF06426206/01/200819,482.590.0011/03/20058413.0024,985.00Balloon478.43EAF06426406/01/200870,054.160.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon2,800.00EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,600.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF06415606/01/200878,539.520.0001/21/200521512.5085,500.00Balloon936.00EAF06422306/01/200897,092.320.0005/02/200517412.00101,650.00Balloon1,300.00EAF06422606/01/200813,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/2008248,502.890.0010/05/20052849.00257,642.00Balloon2,195.23EAF06426206/01/200819,482.590.0011/03/20058413.0024,985.00Balloon478.43EAF06426206/01/200870,054.160.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,100.00EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF06422306/01/200897,092.320.0005/02/200517412.00101,650.00Balloon1,300.00EAF06422606/01/200813,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/2008248,502.890.0010/05/20052849.00257,642.00Balloon2,195.22EAF06425806/01/200819,482.590.0011/03/20058413.0024,985.00Balloon478.43EAF06426206/01/200870,054.160.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,100.00EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF06422606/01/200813,170.980.0005/25/20058411.5024,510.00Regular448.57EAF06425406/01/2008248,502.890.0010/05/20052849.00257,642.00Balloon2,195.23EAF06425806/01/200819,482.590.0011/03/20058413.0024,985.00Balloon478.43EAF06426206/01/200870,054.160.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,000.00EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF06425406/01/2008248,502.890.0010/05/20052849.00257,642.00Balloon2,195.23EAF06425806/01/200819,482.590.0011/03/20058413.0024,985.00Balloon478.43EAF06426206/01/200870,054.160.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,100.00EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF06425806/01/200819,482.590.0011/03/20058413.0024,985.00Balloon478.43EAF06426206/01/200870,054.160.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,100.00EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF06426206/01/200870,054.160.0012/01/200512213.0077,235.00Balloon1,200.00EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,100.00EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF06426406/01/2008105,195.790.0012/08/200534212.00101,080.00Balloon1,100.00EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF06426606/01/2008209,309.060.0012/15/200517913.00211,375.00Balloon2,800.00EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF06427606/01/200887,729.260.0002/08/200611412.0097,280.00Balloon1,500.00EAF06428706/01/2008125,622.640.0003/09/200625412.00125,677.50Balloon1,400.00
EAF064287 06/01/2008 125,622.64 0.00 03/09/2006 254 12.00 125,677.50 Balloon 1,400.00
EAF064296 06/01/2008 123,721.93 0.00 03/16/2006 218 9.50 132,715.00 Balloon 1,280.10
EAF064297 06/01/2008 49,901.41 0.00 03/23/2006 96 12.00 59,670.00 Regular 994.6'
EAF064301 06/01/2008 144,938.57 0.00 03/29/2006 240 11.00 146,250.00 Balloon 1,548.20
EAF064302 06/01/2008 131,996.89 0.00 04/03/2006 222 9.75 138,157.50 Balloon 1,345.7'
EAF064305 06/01/2008 147,175.68 0.00 04/19/2006 276 12.00 146,250.00 Balloon 1,600.00
EAF064312 06/01/2008 197,639.84 0.00 04/27/2006 194 10.00 209,720.00 Balloon 2,184.29
EAF064313 06/01/2008 104,640.31 0.00 04/28/2006 180 10.00 112,125.00 Balloon 1,204.90
EAF064314 06/01/2008 75,968.88 0.00 05/01/2006 132 10.00 88,690.00 Balloon 1,110.39
EAF064322 06/01/2008 136,886.07 0.00 05/25/2006 192 10.00 145,040.00 Balloon 1,516.98
EAF064331 03/01/2008 98,411.04 3,065.94 06/29/2006 162 12.50 103,800.00 Balloon 1,325.00
EAF064333 06/01/2008 91,197.38 0.00 07/19/2006 203 10.25 123,975.00 Balloon 1,288.10
EAF064338 06/01/2008 125,203.95 0.00 08/09/2006 240 10.25 128,968.00 Balloon 1,266.0

Loan	Paid Thru		Interest	Funding	No.		Original	.	Monthly	
Number	Date	Loan Balance	Due	Date	Pmts	Rate	Balance	Loan Type	Payment	
EAF064364	06/01/2008 06/01/2008	103,283.75	$\begin{array}{c} 0.00\\ 0.00\end{array}$	11/17/2006	96 206	10.25	122,760.00	Balloon	1,879.06 1,262.03	
EAF064367		136,067.24 133,978.32		12/08/2006	296 255	10.00	271,260.00	Balloon		
EAF064370	06/01/2008		0.00	12/15/2006	255	10.00	136,818.00	Balloon	1,296.35	
EAF064381	06/01/2008	167,256.48	0.00	01/30/2007	131	10.00	180,279.00	Balloon	2,266.56	
EAF064413	06/01/2008	59,374.44	0.00	04/04/2007	64	12.50	70,200.00	Regular	1,510.00	
EAF064422	06/01/2008	129,514.75	0.00	04/19/2007	187	10.00	133,650.00	Balloon	1,416.29	
EAF064426	06/01/2008	132,047.16	0.00	04/27/2007	256	12.50	270,000.00	Balloon	1,502.97	
EAF064436	06/01/2008	123,046.71	0.00	05/04/2007	161	12.50	126,900.00	Balloon	1,625.00	
EAF064434	06/01/2008	146,701.40	0.00	05/04/2007	157	13.00	151,300.00	Balloon	2,000.00	
EAF064447	06/01/2008	152,619.16	0.00	05/29/2007	174	13.00	156,500.00	Balloon	2,000.00	
EAF064473	06/01/2008	117,712.61	0.00	06/28/2007	235	13.00	119,000.00	Balloon	1,400.00	
EAF064474	06/01/2008	75,569.69	0.00	06/28/2007	98	13.00	81,000.00	Balloon	1,345.00	
EAF064477	06/01/2008	112,828.63	0.00	07/09/2007	225	10.00	114,642.00	Balloon	1,129.99	
EAF064484	06/01/2008	136,860.87	0.00	07/17/2007	272	10.00	138,600.00	Balloon	1,289.98	
EAF064489	06/01/2008	142,646.38	0.00	08/08/2007	312	10.00	143,550.00	Balloon	1,293.35	
EAF064488	06/01/2008	148,622.84	0.00	08/08/2007	324	13.00	149,100.00	Balloon	1,666.01	
EAF064496	06/01/2008	223,515.48	0.00	08/17/2007	212	10.00	227,808.00	Balloon	2,293.19	
EAF064500	07/01/2008	0.00	0.00	08/23/2007	172	13.00	100,900.00	Balloon	1,400.00	
EAF064503	06/01/2008	115,200.44	0.00	08/28/2007	194	10.00	119,295.00	Balloon	1,242.49	
EAF064511	06/01/2008	66,915.25	0.00	09/21/2007	309	14.00	67,100.00	Balloon	805.00	
EAF064512	06/01/2008	150,392.07	0.00	09/27/2007	285	10.00	151,470.00	Balloon	1,393.11	
EAF064518	06/01/2008	166,875.52	0.00	10/17/2007	180	13.00	169,100.00	Balloon	2,139.52	
EAF064536	06/01/2008	158,029.70	0.00	11/29/2007	180	12.00	160,000.00	Balloon	1,920.27	
EAF064543	06/01/2008	86,888.66	0.00	12/14/2007	120	10.00	89,100.00	Balloon	1,177.46	
EAF064542	06/01/2008	61,365.84	0.00	12/14/2007	99	14.00	63,100.00	Balloon	1,075.00	
EAF064546	06/01/2008	104,062.85	0.00	12/19/2007	300	13.00	104,300.00	Balloon	1,176.33	
EAF064549	06/01/2008	136,197.00	0.00	12/21/2007	240	10.00	137,115.00	Balloon	1,323.19	
EAF064551	06/01/2008	64,985.46	0.00	12/26/2007	60	13.00	69,200.00	Regular	1,574.51	
EAF064558	07/01/2008	0.00	0.00	01/17/2008	120	13.00	80,000.00	Balloon	1,194.49	
EAF064566	06/01/2008	90,596.38	0.00	02/11/2008	146	10.00	91,575.00	Balloon	1,086.63 1	3/1/2013
EAF064567	06/01/2008	139,215.02	0.00	02/13/2008	282	10.00	139,590.00	Balloon	1,287.21 1	3/1/2013
EAF064568	06/01/2008	150,811.75	0.00	02/13/2008	231	10.00	151,470.00	Balloon	1,479.85 1	3/1/2013
EAF064570	06/01/2008	178,511.52	0.00	03/07/2008	120	10.00	180,279.00	Balloon	2,382.40 2	4/1/2013
EAF064573	06/01/2008	109,346.50	0.00	03/13/2008	142	13.00	110.000.00	Balloon	1,516.66 1	4/1/2013
LAI 004373	00/01/2008	107,540.50	0.00	03/13/2000	142	15.00	110,000.00	Danoon	1,510.00 1	4/1/2013

Loan Number	Paid Thru Date	Loan Balance	Interest Due	Funding Date	No. Pmts	Rate	Original Balance	Loan Type	Monthly Payment		
EAF064584	06/01/2008	29,722.78	0.00	04/16/2008	72	13.00	30,000.00	Regular	602.22	1	5/1/2014
EAF064593	06/01/2008	69,600.00	0.00	05/12/2008	299	12.50	69,600.00	Balloon	758.89	1	6/1/2013
EAF064594	06/01/2008	45,800.00	0.00	05/12/2008	60	13.00	45,800.00	Regular	1,030.41	1	6/1/2013
EAF064595	06/01/2008	148,500.00	0.00	05/13/2008	166	10.00	148,500.00	Balloon	1,651.63	1	6/1/2013
EAF064600	06/01/2008	59,400.00	0.00	06/06/2008	56	10.00	59,400.00	Regular	1,331.73	1	3/1/2013
EAF064601	06/01/2008	130,000.00	0.00	06/27/2008	180	13.00	130,000.00	Balloon	1,644.81	1	7/1/2013
EAF064602	06/01/2008	127,000.00	0.00	06/27/2008	240	13.00	127,000.00	Balloon	1,487.90	1	7/1/2013
EAF064608		145,530.00	0.00	06/30/2008	276	10.00	145,530.00	Balloon	1,349.33	1	7/1/2013

Total: Florida

7,812,514.88

TOTAL MEDALLIONS

26,572,600.88

Schedules A-1 & A-2

SCHEDULE OF ACQUIRED LOANS TO BE SOLD AT 100.5% OF THEIR UNPAID PRINCIPAL BALANCE

AND

SCHEDULE OF ACQUIRED LOANS TO BE SOLD AT 99.5% OF THEIR UNPAID PRINCIPAL BALANCE

Loan Number	Paid Thru Date	Loan Balance	Interest Due	Funding Date	No. Pmts	Rate	Original Balance	Loan Type	Monthly Payment
EAF053915	06/01/2008	239,895.68	0.00	12/15/2003	144	4.00	349,500.00	Balloon	3,059.97
EAF053944	06/01/2008	242,100.85	0.00	01/30/2004	144	4.00	349,500.00	Balloon	3,059.97
EAF053945	06/01/2008	242,100.85	0.00	01/30/2004	144	4.00	349,500.00	Balloon	3,059.97
EAF053969	06/01/2008	287,681.48	0.00	02/27/2004	144	4.00	407,750.00	Balloon	3,569.97
THE LOANS LISTED ABOVE WILL BE SOLD AT 100.5% - Schedule A-1									
THE LOAN LIST	ED BELOW WI	LL BE SOLD AT	r 99.5% - So	chedule A-2					
EAF053946	06/01/2008	1,092,128.20	2,500.00	01/30/2004	240	4.56	1,449,000.00	Balloon	8,980.52
Note: EAF 053946 adjusts quarterly to the 90 day LIBOR rate plus 150 basis points. On the quarterly change date of July 1,2008 this loan required a monthly payment of \$6,890.49 and the interest rate commencing on July 1, 2008 was reduced to 4.18%									

Schedule B SCHEDULE OF SPECIAL DIVERSIFIED LOANS TO BE ACQUIRED PRIOR TO CLOSING

Loan Number	Paid Thru Date	Loan Balance	Interest Due	Funding Date	No. Pmts	Rate	Original Balance	Loan Type	Monthly Payment
SPECIAL DIVERSIFIED									
EAF024342	06/01/2008	1,989,638.36	0.00	08/22/2006	240	7.75	2,000,000.00	Regular	16,420.73
		1,989,638.36							
Medallion Loan owned by Ameritrans									
ATC001038	06/01/2008	40,753.60	0.00	01/17/2017	170	10.00	42,710.46	Balloon	470.76

Schedule 2.1.2(f)(i) LOANS WITH A LTV IN EXCESS OF 80%

Loan Number	Loan Balance	# of Medallions	Dollar amount in excess of 80%	
EAF054605	348,000.00	3	37,200.00	Chicago
EAF054606	232,000.00	2	24,800.00	Chicago
EAF044555	179,881.00	1	20,000.00	Sommerville, MA

Schedule 2.1.2(f)(ii) METHODOLOGY AND ACCOUNTING PRINCIPLES FOR DETERMINING LTV

	June 2008 Market Value	80% LTV
	value	
Chicago Taxi Loans	129,500	103,600
Miami Taxi Loans	210,000	168,000
Boston, MA	385,000	308,000
Cambridge, MA	425,000	340,000
-		
Sommerville, MA	200,000	160,000

Schedule 3.1.4 EXISTING ENCUMBRANCES TO BE SATISFIED AT OR PRIOR TO CLOSING

Israel Discount Bank of New York as Custodian for Israel Discount Bank of New York, Citibank, NA, Bank Leumi USA, and The US Small Business Administration, per agreements dated September 1993 with Elk Associates Funding Corporation

Schedule 3.1.7(a) LOANS WITH A FINAL SCHEDULED PAYMENT DATE DUE MORE THAN 120 MONTHS AFTER THE DATE OF ITS ORIGINATION

Loan Number	Paid Thru Date	Loan Balance	Interest Due	Funding Date	No. Pmts	Rate	Original Balance	Loan Type	Monthly Payment
EAF054048	06/01/2008	583,312.95	0.00	08/31/2004	180	5.00	711,600.00	Regular	5,627.29
NOTE: This loan changes to Prime +2.00% in October 2008 (9% cap)									
Matures on 10-1-2019									

	Loan	Amount	Remaining Funding
Loan #	Amount	Funded	Obligation
EAF054597	180,000.00	150,000.00	30,000.00

Schedule 3.1.7(f) REQUIRED DISBURSEMENTS TO OBLIGORS

Schedule 3.1.8 COLLATERAL FOR WHICH SELLER DOES NOT HAVE PERFECTED AND VALID FIRST LIEN AND FIRST PRIORITY SECURITY INTEREST

Loan #	Balance at 6.30.08	Comments
		Parri passu participation with Freshstart with Medallion Funding Corp in senior
53996	46,718.65	position
		Parri passu participation with Freshstart with Medallion Funding Corp in senior
53997	46,658.46	position
		Parri passu participation with Freshstart with Medallion Funding Corp in senior
53948	46,207.44	position
54027	47,557.54	Parri passu participation with Freshstart with Medallion Funding Corp in senior position
54065	9.703.93	Parri passu participation with Freshstart with Medallion Funding Corp in senior position
2 1005	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Parri passu participation with Freshstart with Medallion Funding Corp in senior
54086	9,810.17	position

Exhibit A

FORM OF BILL OF SALE

This BILL OF SALE (the "<u>Bill of Sale</u>") is made and given as of ______, 2008 by Elk Associates Funding Corporation ("<u>Seller</u>"), a New York corporation licensed as a Small Business Investment Company by the United States Small Business Administration and a wholly owned subsidiary of Ameritrans Capital Corporation, a Delaware corporation ("<u>Ameritrans</u>"), to Medallion Bank, a Utah industrial bank ("<u>Buyer</u>").

PRELIMINARY STATEMENTS

Ameritrans, Seller and Buyer have entered into a Loan Portfolio Sale and Purchase Agreement dated as of July 16, 2008 (the "Loan Portfolio Purchase Agreement");

Section 2.4.1 of the Loan Portfolio Purchase Agreement contemplates the execution and delivery of this Bill of Sale; and

All capitalized terms used herein and not otherwise defined shall have the meanings attributed to them in the Loan Portfolio Purchase Agreement.

Accordingly, in consideration of the mutual covenants and agreements set forth in the Loan Portfolio Purchase Agreement and hereinafter set forth, and other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the parties hereto do hereby agree as follows:

1. Seller does hereby sell, assign, transfer, convey and deliver to Buyer, and Buyer does hereby accept, all of Seller's right, title and interest in and to the Acquired Loans listed on the Loan Schedule attached hereto as Annex A, together with all right, title and interest of Seller in and to the Loan Documents related to the Acquired Loans, in each case, free and clear of any prior Encumbrances.

2. This Bill of Sale is intended only to effect the transfer of certain assets to be transferred pursuant to the Loan Portfolio Purchase Agreement and shall be governed entirely in accordance with the terms and conditions of the Loan Portfolio Purchase Agreement. Except as expressly set forth in the Loan Portfolio Purchase Agreement, the Loans and the Loan Documents related to the Acquired Loans are being sold without recourse, representation or warranty of any kind or any responsibility or liability whatsoever. Notwithstanding any provision of this Bill of Sale to the contrary, nothing contained in this Bill of Sale shall in any way supersede, modify, replace, amend, waive, expand, enlarge or in any way affect the provisions of the Loan Portfolio Purchase Agreement or any of the obligations or rights and remedies of the parties under the Loan Portfolio Purchase Agreement. 3. This Bill of Sale shall be binding upon, and inure to the benefit of, Seller and Buyer and their respective successors and permitted assigns in accordance with Section 9.4 of the Loan Portfolio Purchase Agreement.

4. This Bill of Sale shall be construed, and the rights and obligations of the parties under this Bill of Sale shall be determined, in accordance with the substantive laws of the State of New York applicable to contracts to be performed entirely within the State of New York as though this Bill of Sale were to be wholly performed within the State of New York and as though the parties hereto were domiciliaries of the State of New York.

IN WITNESS WHEREOF, Seller has caused this Bill of Sale to be executed by its duly authorized representative as of the day and year first above written.

ELK ASSOCIATES FUNDING CORPORATION

By:		
Name:		
Title:		

Annex A to the Bill of Sale

MEDALLION LOAN SCHEDULE

Exhibit B

FORM OF PARTICIPATION EXHIBIT JOINT PARTICIPATION AGREEMENT

AGREEMENT made as of ______ by and between MEDALLION Financial CORP., a New York corporation having its principal office at 437 Madison Avenue, New York, New York 10022 ("MFC"), and ELK ASSOCIATES FUNDING CORP. a New York corporation with principal place of business at 747 Third Avenue, New York, New York 10017 ("ELK").

WHEREAS, MFC is the owner and holder of a certain promissory note described in Schedule A annexed hereto and made a part hereof, made by ______ (the "Borrower") to the order of MFC, in the original principal amount of \$ ______ (the "Note"), evidencing a certain loan in that amount (the "Loan") that is secured by a security agreement dated ______, herewith (the "Security Agreement") covering the Borrowers' C ______ (collectively, "Collateral") all as set forth and more particularly described in the "Loan Documents" as such term is defined below.

WHEREAS, The parties represent to each other that neither is making a loan to the other.

WHEREAS, The parties have agreed, subject to the terms and conditions set forth in this Agreement, that MFC shall have the right and obligation to service the Loan and to enforce the terms of the Loan Documents, as such term is defined below.

NOW THEREFORE, in consideration of the foregoing and for other good and valuable consideration and with the intention of hereby being legally bound, the parties hereby agree as follows:

1. <u>Definitions</u>

(a) "Advances" shall mean all loans, advances, guaranties, extensions of credit, accommodations or any other extensions of credit made or to be made by MFC to or on behalf of Borrower pursuant to the Agreements and all other items chargeable to Borrower's account pursuant thereto.

(b) "Agreements" shall mean the written documents between MFC and Borrower as described in Schedule "A" below (and all related documents), copies of which have been delivered by MFC to ELK.

(c) "Collateral shall mean all Chicago taxicab licenses and representative medallions and other property, security, interests and/or guaranties received by MFC pursuant to the agreements or otherwise acquired by MFC in connection with the Transactions.

(d) "Collections" shall mean all monies received by MFC as principal, interest, discount, bonus or commissions on account of the Transactions in whatever manner from whatever source or as proceeds of the Collateral or from any Guarantor.

(e) "Maximum Investment" in reference to ELK shall mean the principal sum of

(f) "ELK's Investment" shall mean ELK's initial contribution hereunder on account of its Participation, in the amount of \S ______, plus all additional sums paid by ELK to MFC on account of its Participation hereunder and less all principal payments received by ELK in connection with the Transactions.

(g) ELK's "Participation Interest" or "ELK's Participation Percentage" shall mean at any point in time ELK's fully subordinated fractional interest in the Transactions, Advances, Collateral and Collections.

(h) "Transactions" shall mean all actions taken and to be taken by MFC, Borrower and/or any other party in connection with the Agreements.

2. <u>Participation</u>

(a) Effective as of the date of this Agreement, the respective or pro-rata interests of the parties in the Loan, or so much thereof as may be advanced, the Loan Documents and the Collateral, shall be as follows:

MFC (hereinafter referred to at times as "MFC's Senior Participation Percentage")

ELK (hereinafter referred to at times as "ELK' Junior Participation Percentage")

(b) ELK shall own a fully subordinated fractional interest in the Loan, in its respective _____ (_____%) percent Participation Percentage and in all documents, instruments and collateral issued by the Borrower or the Guarantors (collectively referred to as "Obligors") to MFC. ELK's interest in the Loan and in all documents, instruments and collateral

issued by the Borrower or the Guarantors is fully and wholly subject and subordinate to the interests of MFC in the said Loan and in all documents, instruments and collateral issued by the Borrower or the Guarantors as detailed herein. Further ELK's interest in all payments made thereon and in any recoveries or distributions in connection therewith is fully and wholly subject and subordinate to the interests of MFC therein as detailed herein.

3. Interest

(a) ELK shall earn interest, subject to its subordination of its interest to MFC as set forth herein, on its (ELK's) Investment according to its (ELK's) Participation Percentage at the rate ______ per annum during the term of the Loan, less a collection fee which may be deducted by MFC equal to ______ basis points. Interest shall be calculated based upon a 360 day year for the actual number of days elapsed. Such interest may be paid, subject to the terms and conditions herein contained, within ten (10) days of payment due date set forth in the Borrower's Promissory Note executed and delivered to MFC in connection with the Loan.

(b) Upon receipt of funds from Borrower, whether representing a monthly payment(s) or portion thereof, or any other payment from the Borrower, MFC shall, subject to the disbursement order set forth herein, within five (5) business days thereof, tender to ELK payment(s) consisting of interest on its ELK's Investment at the rate set forth above. Elk's portion of any principal received from routine monthly payments paid prior to any default by the Borrower shall also be remitted to Elk.

4. <u>Priority of Payments</u>

(a) (i) Payments received by MFC from Borrower with respect to the Loan, prior to a default by the Borrower and prior to MFC taking possession of any collateral pertaining to the Loan, shall be divided by MFC on a pro-rata basis pursuant to the respective Participation Ratios heretofore set forth herein. Such payments shall first be paid to MFC. In the event the Loan becomes delinquent, MFC shall apply payments pursuant to the terms of this Agreement. Notwithstanding the foregoing, MFC shall first be paid all additional amounts which it has advanced where the Borrower has failed to make payments to MFC and expenses reasonably

3

incurred by MFC in connection with the enforcement of the Agreements and/or protection of the Collateral and the secured position therein.

(ii) Following application of payments as set forth in sub paragraph (a) (i) hereof, MFC shall remit payment to ELK representing its share of such payment(s) based upon its respective ______ per cent Participation Percentage. Again, in the event the Loan becomes delinquent, MFC shall apply payments pursuant to the terms of this Agreement. Notwithstanding the foregoing, MFC shall first be paid all additional amounts which it has advanced to include monies previously advanced to ELK where the Borrower has failed to make payments to MFC and expenses reasonably incurred by MFC in connection with the enforcement of the Agreements and/or protection of the Collateral and the secured position therein.

(b) In the event of a default under the Loan and MFC taking possession of any Collateral pertaining to the Loan and upon subsequent resale thereof, the priority of payments shall be as follows:

1. First, MFC shall be paid all additional amounts which either has advanced for expenses reasonably incurred in connection with the enforcement of the Agreements and/or protection of the Collateral and the secured position therein;

- 2. Then to the payment of the outstanding principal due to MFC on a pro-rata basis pursuant to the Participation Percentages heretofore set forth herein;
- 3. Then to the payment of accrued interest to MFC at the face rate specified

in the Promissory Note (but not any default rate of interest) on a pro-rata basis

pursuant to the Participation Percentages heretofore set herein;

- 4. Then to MFC on a pro-rata basis for any late charges collected, but only to MFC's pro-rata share of such late charges collected;
- 5. Then, the balance, if any, to ELK for its pro-rata share of principal and accrued interest on the Loan at the face rate set forth in the Promissory Note (but not any default rate of interest);
- 6. Then, the balance, if any, to Elk for its pro-rata share of any late charges;

7. Then, the balance, if any to MFC and ELK on a pro rata basis according to their respective Participation Percentage for all other sums for prepayment penalties, default interest, or other revenue collected arising out of the liquidation of the Collateral.

(c) ELK's interest in the Loan, the Collateral and the Collections is in all respects junior and subordinate to that of MFC, and accordingly MFC shall, following default and repossession of the Collateral, first be repaid in full, its' respective interests in the Loan in the priority and manner stated in Paragraph 4 (b) items 1 through 7 above, out of the Collateral and the Collections prior to any repayment of all or any part of ELK's Investment and the interest equivalent thereon, other than as may be set forth in said Paragraph 4 (b) items 1 through 7 above, or as may otherwise be specifically set forth elsewhere in this Agreement.

5. <u>Procedure</u>

(a) All of the Agreements shall be in the name of MFC as payee, obligee, or otherwise, however all payments shall be made to MFC and MFC shall hold same in its possession as trustee for the for the benefit of ELK to the extent of its Participation. MFC shall not apply any amount received from or on behalf of Borrower to any obligation other than the Accounts.

(b) MFC will keep accurate records of all Advances, Collateral, and Collections apart from its records of the other assets, appropriately marked to show the interest of ELK therein. In recording MFC's ownership of the Loan on its books and in statements of condition and assets, MFC shall clearly define ELK's interest in the loan and Collateral. ELK or its designees may inspect same upon reasonable notice during normal business hours. MFC will furnish to ELK upon reasonably request, such information it may have concerning the Agreements, the performance thereof, and/or the financial condition and operation of the Borrower. In connection with MFC's responsibilities to clearly define Elk's interest in the Loan and the Collateral, MFC agrees to do the following:

- MFC agrees to add a notice on the first page of the Promissory Note that states: "<u>Notice of Assignment</u>: A 20% subordinate interest of the within Note has been assigned to Elk Associates Funding Corporation as more fully described on the signature page of this Note".
- 2) On the signature page of the Promissory Note, MFC agrees to place the following endorsement on such page which shall read as follows: "PAY TO THE ORDER OF ELK ASSOCIATES FUNDING CORPORATION AN

UNDIVIDED 20% SUBORDINATE INTEREST IN THIS NOTE EQUALING AN INITIAL INTEREST THEREIN OF \$23,400.00., SUBJECT TO THE TERMS AND CONDITIONS OF A CERTAIN JOINT PARTICIPATION AGREEMENT BY AND BETWEEN MEDALLION FUNDING CORPORATION AND ELK ASSOCIATES FUNDING CORPORATION CONCERNING THIS PROMISSORY NOTE, THE LOAN BEING MADE TO MAKER, AND THE COLLATERAL GIVEN TO SECURE REPAYMENT OF THIS PROMISSORY NOTE. Signed:

MEDALLION FINANCIAL CORPORATION

BY:_____

(c) All collections received with MFC in connection with the Transactions shall be applied to Borrower's account with MFC as provided in the Agreements and, thereafter paid to MFC, and, if applicable, then to ELK, as set forth herein. Regular monthly payments collected from the Borrower shall be remitted to Elk within fifteen (15) days after receipt thereof by Medallion. Payments due to Elk arising out of any full or substantial partial repayment of the Loan shall be remitted by MFC to Elk within three business days of receipt by MFC of good and clear funds for such payment.

(d) MFC makes no representation or warranty as to Borrower's creditworthiness or financial condition or the collectability of the Loan or Collateral. ELK has made its own independent evaluation and determination of Borrower's creditworthiness. MFC warrants the sufficiency, legality, and enforceability of the Agreements including, but not limited to, a perfected first security interest in the Collateral, according to their terms, except as limited by bankruptcy laws and limitations and equitable remedies generally, and that ELK may purchase its interest hereunder without necessity of registration under the Securities Laws. MFC agrees to supply Elk with copies of all loan applications, credit reports, tax returns, and due diligence documentation or other background information in MFC's possession pertaining to the Borrower or any guarantors of the Loan related to this Loan.

(e) MFC shall bear all costs and expenses of managing and servicing the Transactions, and in consideration thereof, shall earn an arm's length fee equal to the difference in the interest collected from the Borrower and that paid to ELK, provided however that such fee shall not exceed a total of 25 basis points from the actual interest that may be collected by MFC from time to time, except as provided herein.

(f) MFC shall have the right to purchase ELK'S interest in the Loan by giving ELK twenty (20) days written notice of its intent to purchase and upon payment of ELK'S Investment, plus any interest accrued thereon (to date of payment) and late charges, if any, ELK shall release and assign its Participation Interest to MFC.

(g) MFC agrees to assign of record Elk's subordinate participation interest in the Loan by amending any UCC Financing Statements that MFC may record against the Borrower to indicate that MFC has sold and assigned a 20% subordinate participation interest in the collateral securing the Loan. Said UCC 3 Amendments shall be filed in such recorder's offices as to give Elk a perfected filed security interest of record in its Participation Interest in the Loan.

(h) In the event the Loan is in default beyond a period of ninety (90) days from the due date of any payment, then in such event, Elk is hereby given the right, but not the obligation, at any time thereafter while the Loan remains in default to purchase MFC's senior interest in the Loan for the amount of MFC's principal, accrued interest at the face rate then in effect to the date of payoff, and any MFC's portion of any late charges that have accrued. MFC shall not be entitled to payment of any prepayment penalty or default interest in calculating the amount at which ELK will pay MFC to purchase MFC's interest therein.

6. <u>Management</u>

(a) MFC shall have the right and obligation to manage, perform, and enforce the terms of the Agreements and to exercise all privileges and rights exercisable or enforceable thereunder, according to MFC's customary procedures with similar borrowers and in the exercise of its prudent business judgment.

(b) MFC shall not without the prior written consent of ELK modify or amend any provision of the Agreements or waive or release any mutual rights thereunder.

7

(c) ELK shall have no obligation to make any payment to MFC under Paragraph 5(c) to the extent that MFC has made Advances in excess of the maximum amount available under the Agreements.

(d) MFC shall use reasonable prudence and judgment in the servicing of the Transactions, carrying out the terms of the Agreements and the collection of amounts due from Borrower which servicing shall include delivering copies of all documents and reports obtained by MFC relating to the Borrowers, any Guarantors and Collateral. MFC agrees, in servicing the Transactions, to conform to applicable regulations and laws and to enforce such Agreements in a commercially reasonable manner.

(e) MFC shall not hypothecate, pledge, assign, or otherwise encumber ELK'S Participation Interest in the Loan.

(f) ELK may sell, pledge, assign, or sub-participate or otherwise transfer any or all of its right in the Loan to any other third party entity on the condition that it first provide MFC, at least (10) days prior to the contemplated closing of any such transaction, written details of the subject transaction including, without limitation, the name of the prospective purchasing entity and the terms of the transaction and, in conjunction therewith, MFC shall have a right of first refusal with regard to such transaction on the same terms and conditions set forth in the description of the contemplated transaction. (Notwithstanding the foregoing, MFC shall not have a right of first refusal with respect to any pledge of Elk's Participation Interest given by ELK to its banks or the US Small Business Administration pursuant to any financing documents now or hereafter existing between ELK and said parties). In the event MFC desires to exercise such right of first refusal, it shall notify ELK, in writing, prior to the expiration of the aforestated ten (10) day period, of its intention to exercise such right and shall consummate the transaction within fifteen (15) days thereof. In the event MFC fails to timely notify ELK of its intention to exercise its right of first refusal hereunder, or in the event MFC fails to timely consummate the subject transaction within the time parameters set forth in this sub-paragraph, MFC shall have been deemed to have waived its rights of first refusal under this sub-paragraph.

(g) MFC shall be entitled to pledge or grant a security interest in its Participation Interest in the Loan to any institutional lender as part of a financing plan to fund MFC's loan portfolio without obtaining ELK's consent thereto. MFC shall also have the right to assign, or otherwise transfer any or all of its rights in the Loan to any subsidiary, parent or affiliate of MFC or business entity owned or controlled by or under Medallion Financial Corp., without obtaining ELK's consent thereto. In the event MFC desires to sell or assign or otherwise transfer any or all of its right in the Loan to any third party nonaffiliated entity, it shall be on condition that it first provide ELK, at least ten (10) days prior to the contemplated closing of any such transaction, written details of the subject transaction and, in conjunction therewith, ELK shall have a right of first refusal with regard to such transaction. In the event ELK desires to exercise such right of first refusal, it shall notify MFC, in writing, prior to the expiration of the aforestated ten (10) day period, of its intention to exercise such right and shall consummate the transaction within fifteen (15) days thereof. In the event ELK fails to timely notify MFC of its intention to exercise its right of first refusal hereunder, or in the event ELK fails to timely consummate the subject transaction within the time parameters set forth in this sub-paragraph, ELK shall have been deemed to have waived its rights of first refusal under this sub-paragraph.

(h) As previously stated herein, the parties acknowledge that MFC will receive as a fee for its services and management hereunder interest payment from the Borrowers at the rate set forth on the Promissory Note, and ELK agrees that MFC may also deduct 25 basis points from the interest otherwise due ELK on its Participation Interest as a service and management fee. The fee earned by MFC is an arm's length transaction.

7. <u>Representations and Warranties</u>

(a) MFC hereby represents and warrants that it holds at present the Agreements true copies ELK, which confirms that it has examined the Agreements and is fully familiar with the terms and provisions thereof.

(b) MFC represents and warrants that it has the right and authority to enter into this Agreement as of the date hereof. ELK is receiving its Participation Interest in the Loan free and clear of all liens and encumbrances. MFC represents that as of the date ELK pays for its participation, the outstanding principal balance of the Loan is \$585,000.00 and Borrower is not in default under the Agreements.

(c) Except as expressly set forth in this Agreement and the documents delivered in connection herewith, no party makes any express or implied representation or warranty with respect to the Loan, the collect ability of the indebtedness evidenced by the Notes, the credit

9

worthiness of the Borrower, the enforceability of any of the Agreements or the value of the collateral.

8. <u>Default</u>

(a) If a Borrower remains in default under any Loan for thirty (30) days, then within three (3) business days thereafter, MFC will give ELK written notice of such default and simultaneously deliver copies of any written notice given to such Borrower. MFC hereby agrees that in the event of a default by the Borrower which remains uncured for seventy five (75) days, it shall endeavor to commence foreclosure proceedings against the subject Borrower and all Collateral security the Loan. in any event, if such default remains uncured for a period of one hundred twenty (120) days, MFC must commence foreclosure proceedings unless ELK has previously agreed, in writing, to a forbearance regarding same. In the event MFC fails for any reason to enforce the Agreements pertaining to such Loan, then ELK shall have the right, in addition to any other rights it may have pursuant to the terms of both the Agreements and this agreement, or at law or equity, on fifteen (15) days prior written notice to MFC, to take whatever steps it deems necessary and appropriate to protect its Participation Interest in such Loan. ELK's rights with respect thereto shall include the independent right to commence a foreclosure proceeding, or appropriate litigation in its own name, as a subordinate holder of a security interest in the Collateral given by Borrower to secure the Loan, and thereafter ELK shall have the right to conduct the foreclosure sale or litigation, hire attorneys with respect thereto, disburse expenses and any taxes, parking tickets, violations, etc necessary to secure the Collateral, and pay obligations of the Borrower so that the Collateral can be sold and transferred at the City of Chicago Department of Consumer Services to a new purchaser thereof. All such disbursements paid by Elk to secure the Collateral, pay attorneys fees, and to pay and discharge taxes and other liabilities of Borrower necessary to permit the completion of a foreclosure sale shall be repaid to ELK out of the proceeds of the sale in the same order of priority to ELK as if MFC had paid for same and was seeking reimbursement thereof according to the terms and conditions set forth for reimbursement of such items as described in Paragraph 4 (a) (i) or 4 (b) (i), or as may otherwise be specifically set forth elsewhere in this Agreement. In the event Elk exercises its rights hereunder and gives MFC notice thereof as required by the terms of this paragraph, upon receipt of such notice, MFC shall promptly deliver to ELK such Agreements, including, but not limited

to, Notes and Security Agreements and other Agreements and records relating to the Loan, together with all necessary and proper assignments and documents of authority, that ELK may request for the purpose of protecting its Participation interest.

9. <u>General</u>

(a) All notices provided for herein shall be in writing and hand-delivered, telecopied or mailed by United States first class mail to the respective parties at the addresses first set forth above, or at such other addresses as either of them shall have specified to the other in writing.

(b) This Participation Agreement: (i) shall be binding upon and inure to the benefit of the parties hereunder and their respective legal representatives, successors and assigns; (ii) shall be governed, construed, and interpreted in all respects in accordance with the laws of the State of New York without regard to conflicts of law rules; (iii) may not be modified, amended, terminated or otherwise changed orally or by any course of dealing or in any manner except by an agreement in writing signed by the duly authorized officer of the party to be changed; and (iv) shall not create a partnership or joint venture.

(c) Nothing herein shall be deemed to create a joint venture or partnership between the parties hereto, nor shall the entering into of this Agreement be construed as a loan by either party to the other.

(d) In the event of (1) a filing of MFC of a petition under any provision of the bankruptcy law or of an assignment for the benefit of creditors, or (2) the filing against MFC of a petition under any bankruptcy law that is not dismissed within sixty (60) days, or (3) the sale of the shares and/or assets of MFC in conjunction with MFC's dissolution and liquidation; then ELK shall have the right (but not the obligation) to succeed, at its option, to all rights and duties that MFC may have regarding the holding and enforcing of the terms of the Agreements and the servicing of the Loan, and ELK shall have the right (but not the obligation) to exercise all of the powers granted under this Agreement to MFC and shall have the option to designate itself to exercise such powers in a manner consistent with this Agreement and notifies MFC of such designation, MFC shall promptly deliver to ELK, Agreements, including the Notes, Security Agreements and all others Loan Documents and records relating to the Loan, together with all necessary and proper assignments and documents of authority, and ELK shall also be entitled to

receive from MFC its share of reasonable expenses incurred and disbursements made by it or its designee in holding or enforcing the terms of the Agreements or servicing the Loan.

(e) Each of the parties hereby expressly submits and consents in advance to the jurisdiction of the Supreme Court of the State of New York in the County of New York in any action or proceeding commenced by the other in such court with reference to any claim or dispute between the parties pertaining directly or indirectly to this Agreement or to any matter arising therefrom. Each of the parties hereby waives the right to trial by jury in any such action or proceeding.

(f) MFC represents and agrees that the making and performance of this Agreement is within its corporate powers and has been duly authorized by all necessary corporate action; that it is in compliance with all laws and regulations by which it is bound and does not conflict with any other agreements by which it is bound; this Agreement has been duly executed by it and constitutes the legal, valid and binding obligation of MFC enforceable in accordance with its terms.

(g) Effective date: The parties acknowledge that the effective date of this agreement is and that there is no interest obligation until the loan if funded.

MEDALLION FINANCIAL CORP.

ELK ASSOCIATES FUNDING CORP.

By_____

By_____

PARTICIPATION ONLY ELK ASSOCIATES FUNDING CORP. LOAN FUNDING REQUEST

CUSTOMER:		
FUNDING DATE:		
TOTAL LOAN:	<u>\$</u>	
<u>CHECK #</u>	VENDOR NAME	<u>AMOUNT</u>
	TOTAL DISBURSEMENT	\$

APPROVED POSTED BY	
FUNDED BY	

LOAN NUMBER